# PROPERTIES OF CONCRETE CONTAINING RECYCLED AGGREGATE AND WASTE GLASS POWDER

MD. LUTFAR RAHMAN TALUKDER RANA

# **M.Sc. ENGINEERING THESIS**



# DEPARTMENT OF CIVIL ENGINEERING MILITARY INSTITUTE OF SCIENCE AND TECHNOLOGY DHAKA, BANGLADESH

**SEPTEMBER 2023** 

# PROPERTIES OF CONCRETE CONTAINING RECYCLED AGGREGATE AND WASTE GLASS POWDER

# MD. LUTFAR RAHMAN TALUKDER RANA (SN. 0419110019)

A Thesis Submitted in Partial Fulfillment of the Requirements for the Degree of Master of Science in Civil Engineering



DEPARTMENT OF CIVIL ENGINEERING MILITARY INSTITUTE OF SCIENCE AND TECHNOLOGY DHAKA, BANGLADESH

SEPTEMBER 2023

### PROPERTIES OF CONCRETE CONTAINING RECYCLED AGGREGATE AND WASTE GLASS POWDER

# M.Sc. Engineering Thesis By MD. LUTFAR RAHMAN TALUKDER RANA (SN.0419110019)

Approved as to style and content by the Board of Examination on 14 September 2023:

Lt Col Khondaker Sakil Ahmed, PhD, PEng, CEng Associate Professor Department of Civil Engineering MIST, Dhaka

Chairman (Supervisor)

Dr. Md. Ataur Rahman Professor Department of Civil Engineering KUET, Dhaka

Maj Md. Soebur Rahman, PhD, PEng, CEng Associate Professor Department of Civil Engineering MIST, Dhaka Member (External)

Member (Internal)

Colonel Mir Sarwar Hossain Chowdhury, afwc, psc Acting Head Department of Civil Engineering MIST, Dhaka Member (Ex-officio)

Department of Civil Engineering MIST, Dhaka

# PROPERTIES OF CONCRETE CONTAINING RECYCLED AGGREGATE AND WASTE GLASS POWDER

## DECLARATION

I hereby declare that the study reported in this thesis entitled as above is my/our own original work and has not been submitted before anywhere for any degree or other purposes. Further, I certify that the intellectual content of this thesis is the product of my own work and that all the assistance received in preparing this thesis and sources have been acknowledged and or cited in the reference section.

## MD. LUTFAR RAHMAN TALUKDER RANA

#### ACKNOWLEDGEMENT

First and foremost, all praise and profound gratitude goes to the almighty Allah, who is the most beneficent and the most merciful, for allowing the author a great opportunity and the ability to bring this effort to fruition safely and peacefully.

The author conveys his heartiest gratitude to his supervisor, Lt Col Khondaker Sakil Ahmed, Ph.D., PEng., CEng. Associate Professor, CE Dept., MIST for his systematic support and supervision of the dissertation. His guidance and suggestions worked as inspiration to unlock different new avenues for research. The author would like to thank Dr. Ataur Rahman, Professor, CE Dept., KUET for the useful comments, remarks, and valuable suggestions.

The author is grateful for the technical support provided by the Center for Advisory & Testing Services of the Civil Engineering Department (CATS-CE) of MIST, Dhaka, Bangladesh. The author would also like to acknowledge the contribution of lab technicians of the Concrete and SM lab of the Civil Engineering Department of MIST.

#### ABSTRACT

#### Properties of Concrete Containing Recycled Aggregate and Waste Glass Powder

The coarse aggregate and fine aggregate are replaced by the RBA and WG, respectively. A mix design of a total of fifteen different mixtures is planned with a comprehensive test matrix. Fundamental tests of concrete such as density, gradations, air content, slump, as well as the mechanical properties i.e. compressive strength, splitting tensile strength, and flexural strength of RBA-based WG concrete at different ages are investigated following ASTM standards. Subsequently, a non-destructive test method ultrasonic pulse velocity is also examined to understand the pulse conductivity through produced concrete. For further investigation, a microstructural analysis is also performed using scanning electron microscopy. It is observed that the compressive strength diminishes with a higher amount of reclaimed brick aggregate. In contrast, a 12.5% higher strength can be achieved with the addition of 20% glass content though the strength falls if further glass is added as fine content. Similar trends are observed for the split tensile and flexural strength properties of this RCG. In the SEM, it is observed that up to 20% glass content RCG becomes denser formation with lesser voids; however, the larger voids and fissures are observed as glass content increases.

The research has been extended to investigate the flexural behaviour of the RCG in the reinforced concrete beam. Load-deformation of the beam and stress-strain response of the used rebar's were recorded by a standard data acquisition system including LVDT, strain gauges with a 20-channel data logger. The flexural capacity of RCG-made beams is very much comparable with control beams as the deviation is only in a range of 1%-8% for adding glass content up to 20% replacement. The experimental results are compared with the established codes such as ACI-318-14. The experimental value is near to the theoretical value calculated through the ACI-318-14 code. The outcome of this study may contribute to the application of RBA and WG as ingredients of normal and reinforced concrete.

# NOTATION

RCG	Recycled Concrete with Glass
RBA	Recycled Brick Aggregate
RBCA	Recycled Brick Coarse Aggregate
WG	Waste Glass
RGFA	Recycled Glass Fine Aggregate
RG	Recycled Glass
EPD	Environmental Prottection Department
MSW	Municipal Solid Waste
RGBC	Recycled Glass Brick Concrete
WGP	Waste Glass Particle
RC	Reinforced Concrete
SEM	Scanning Electron Microscopy
ACI	American Concrete Institute
ASTM	American Society for Testing Materials
LVDT	Linear Variable Differential Transformer
ASR	Alkali Silica Reaction
NCA	Natural Coarse Aggregate
UPV	Ultrasonic Pulse Velocity
Pcr	Cracking Load (kN)
Ру	Yeilding Load (kN)
Pu	Ultimate Load (kN)
Mcr	Cracking Moment (kN-m)
Му	Yeilding Moment (kN-m)
Mu	Ultimate Moment (kN-m)

# LIST OF TABLES

Table 2-1:	Chemical composition of Chemical composition of cement, sand & different glasses	8
Table 2-2:	Slump values of concrete containing glass concrete	11
Table 2-3:	Influence of recycled glass on Compressive Strength of glass concrete	15
Table 2-4:	Splitting Tensile Strength of Concrete containing RG as Coarse or Fine Aggregate	21
Table 2-5:	Flexural properties of glass concrete	23
Table 3-1:	Physical properties of aggregate	34
Table 3-2:	Specimen identification description	36
Table 3-3:	The proportions of aggregates for concrete mixture per cubic meter (Cylindrical)	37
Table 3-4:	The proportions of aggregates for concrete mixture per cubic meter (beam)	37
Table 3-5:	Test matrix	38
Table 4-1:	Air content and Slump test result of RCG	50
Table 4-2:	Compressive strength of RCG (7,28,56 days)	54
Table 4-3:	Compressive and Tensile strength of RCG	56
Table 4-4:	Necessary equations from different codes	61
Table 5-1:	Flexural capacity of RCG beam	80
Table 5-2:	Ultimate moment capacity equation (ACI-318-14)	80

# LIST OF FIGURES

Fig. 1.1:	Recovery percentage of waste glass in Hong Kong	2
Fig. 1.2:	Glass Waste Management of U.S.A. (EPA-1960-2018)	2
Fig. 1.3:	Demolished of an old administrative building in Bangladesh	3
Fig. 2.1:	Research statistics of literature on Recycled Glass Concrete	6
Fig. 2.2:	Appearance of natural and glass sand	7
Fig. 2.3:	SEM images of glass adapted with permission from Elsevier	7
Fig. 2.4:	CaO-AL <sub>2</sub> O <sub>3</sub> -SiO <sub>2</sub> ternary diagram of glass powder, FA and GBFS, and Portland cement	9
Fig. 2.5:	Typical slump test of concrete	10
Fig. 2.6:	Slump values of fresh recycled glass concrete	12
Fig. 2.7:	Flow of cement mortar containing different proportions and types of glasses	13
Fig. 2.8:	Air content with glass sand content for different concrete grades and colors of glasses	14
Fig. 2.9:	Influence of Glass as a replacement of sand on the Compressive Strength for different ,mixtures ,ages,colors and sizes	18
Fig. 2.10: Fig. 2.11:	Influence of varying waste glass proportions on the Compressive Strength of cement mortar for different ages, glass color and W/C ratio	20 22
Fig. 2.12:	Flexural strength of RGC with varying sand replacement by	24
Fig. 2.13:	Conductivity of glass concrete for different concrete grades	26
Fig. 2.14:	Expansion behavior of different % of RG as a replacement of Coarse Aggregate in concrete	27
Fig. 2.15:	Fresh and hardened density of concrete with WG	29
Fig. 2.16:	Density of Concrete containing RG as replacement of CA	30
Fig. 3.1:	Core materials of this research (Recycled Brick and Waste Glass)	34
Fig. 3.2:	Particle Size Distribution Curve of WG and Natural Sand	35
Fig. 3.3:	Slump test of Concrete mix and Air Content test of Concrete mix	39
Fig. 3.4:	Mixing, Hammering, Casting and Notation	40
Fig. 3.5:	Strain gauge installation process of beam	40
Fig. 3.6:	Notation of beams and cylinder	41

Fig. 3.8:	Curing process of cylindrical, prism beam and beam specimen	41
Fig. 3.9:	Compression test setup of Cylindrical specimen	42
Fig. 3.10:	Principle and method of wave transmission used in UPV	43
Fig. 3.11:	UPV test of cylindrical specimen	43
Fig. 3.12:	Coating on sample	44
Fig. 3.13:	Three point bending test setup	45
Fig. 3.14:	Four-point bending test setup	46
Fig. 4.1:	Slump of RCG	49
Fig. 4.2:	Air Content of RCG	50
Fig. 4.3:	Compressive strength of RCG (7 days)	52
Fig. 4.4:	Compressive Strength of RCG (28 days)	53
Fig. 4.5:	Compressive strength of RCG (7,28,56 days)	55
Fig. 4.6:	Splitting Tensile Strength of RCG	56
Fig. 4.7:	Compressive and tensile strength (28 days)	57
Fig. 4.8:	Flexural Strength of RCG (28 and 56 days)	58
Fig. 4.9:	Dry density of RCG	59
Fig. 4.10:	Dry density Vs Compressive strength of RCG	59
Fig. 4.11:	Ultrasonic Pulse Velocity of RCG	60
Fig. 4.12:	Relation between compressive strength and UPV	61
Fig. 4.13:	Splitting Tensile strength values according to codes	62
Fig. 4.14:	Flexural strength values according to codes	62
Fig. 4.15:	Ratio between experimental and calculated values of Splitting Tensile strength	63
Fig. 4.16:	Ratio between experimental and calculated values of Flexural strength	63
Fig. 4.17:	Failure pattern of RCG concrete cylinder in tension	64
Fig. 4.18:	Failure pattern of RC30G40 prism beam in flexure	65
Fig. 4.19:	Flexural failure pattern of RCG concrete	65
Fig. 4.20:	SEM images of $RC_{30}G_{100}$ sample at 30µm and 10µm scale	66
Fig. 4.21:	SEM images of RC <sub>30</sub> G <sub>20</sub> sample at 200µm and 10µm	67

Fig. 5.1:	Load VS Deflection behavior of RCG beam	72
Fig. 5.2:	Deflection behavior from LVDT	73
Fig. 5.3:	Strain Gauge data (Load Vs Strain)	74
Fig. 5.4:	Crack of RC <sub>30</sub> G <sub>80</sub> beam Specimen	77
Fig. 5.5:	Crack of RCG beam Specimen	78
Fig. 5.6:	Ultimate moment capacity of RCG beam	81
Fig. 5.7:	Comparison of Ultimate moment capacity of RCG beam with ACI-318 (2014)	81

ACKNOWLEDGEMENT					
ABSTRACT					
NOTAT	ION	iii			
LIST OF	TABLES	iv			
LIST OF	FIGURES	v			
TABLE	OF CONTENTS	viii			
CHAPTI	ER 1 : INTRODUCTION	1			
1.1	Introduction	1			
1.2	Background and Motivation	3			
1.3	Research Objectives	4			
1.4	Scope of the Thesis	4			
1.5	Organization of the Thesis	5			
CHAPTI	ER 2 : LITERATURE REVIEW	6			
2.1	Introduction	6			
2.2	Waste Glass	7			
2.3	Chemical Composition of Glass				
2.4	Fresh Properties of Glass Concrete	9			
	2.4.1 Workability	9			
	2.4.2 Air Content	13			
2.5	Hardened Properties of Concrete	14			
	2.5.1 Compressive Strength	14			
	2.5.2 Splitting Tensile Strength	21			
	2.5.3 Flexural Strength	23			
2.6	Conductivity	25			
2.7	ASR Expansion	26			
2.8	Density of RG Concrete	28			
2.9	Properties of Recycled Brick Aggregate	30			
2.10	Summary	31			
CHAPTI	ER 3 : METHODOLOGY	32			
3.1	Introduction	32			
3.2	Material Collection and Physical Properties Test of Aggregates	32			

	3.3	Mixture Proportions	36
	3.4	Workability Test of Concrete	39
	3.5	Casting and Notation	39
	3.6	Mechanical Properties test of RCG	42
	3.7	Ultrasonic Pulse Velocity Test	43
	3.8	SEM Test	44
	3.9	Flexural Test Setup	45
	3.10	Summary	46
C	HAPTI	ER 4 : FRESH AND HARDENED PROPERTIES OF RCG	47
	4.1	Introduction	47
	4.2	Workability of RCG	47
	4.3	Compressive Strength	52
	4.4	Splitting Tensile Strength	56
	4.5	Flexural Strength	58
	4.6	Hardened Concrete Density	59
	4.7	Ultrasonic Pulse Velocity	61
	4.8	Comparison with the Standard Code of Practices	62
	4.9	Failure Pattern of Concrete	65
	4.10	SEM Images	65
	4.11	Summary	66
	CHAI	PTER 5 : FLEXURAL RESPONSE OF REINFROCED CONCRETE BEAM	68
	5.1	Introduction	68
	5.2	Load – Deflection Characteristics	69
	5.3	Crack Pattern of Beam	76
	5.4	Flexural Strength of Beam	78
	5.5	Summary	79
	CHAI	PTER 6 : CONCLUSIONS AND RECOMMENDATIONS	80
	6.1	Conclusions	80
	6.2	Recommendations for Future Research	81
	REFE	RENCES	82

#### CHAPTER 1 INTRODUCTION

#### **1.1 Introduction**

Various artificial derivations have been created by the construction sector that are effective at enhancing sustainability. Reusing the materials, equipment, and wastes that would otherwise burden the terrain and enclose small pockets of available land reduces the loss of land and the landscape due to relatively less digging. Nowadays, recycling is a common practice because it protects the planet's resources (Terro, 2006). It is estimated that only 85.9 million metric tons of cement are manufactured in the United States alone, as opposed to the staggering 200 million metric tons produced globally (US Geological Survey, 2017). Twenty billion tonnes of concrete are thought to be consumed worldwide per year. We require 14 billion tonnes of natural aggregate to produce this much concrete.

According to a recent estimation, the globe must deal with the issue of disposing of 12 billion tonnes of demolished concrete trash every time buildings and other reinforced concrete designs are destroyed . Around 17.2 billion complexions of burnt bricks are produced every time through around,5000 slip-up kilns across the country. According to reports, the USA produced 11.4 million tons of trash glass in 2017, accounting for 4.2 percent of total external solid waste at that time. In the UK, the Department of Environment, Food, and Rural Affairs (DEFRA) reported that the UK generated 2.4 million tons of waste glass in the time 2017. Only 33 percent of waste glass in the USA is recovered; the remainder is dumped in dumpsters. Another report claimed that 67.6 percent of the glass garbage produced in the UK was collected in 2017.

The statistics on the recycling and disposal of waste glass in Hong Kong from 1997 to 2015 were gathered from the Environmental Protection Department (EPD) report (Environmental Protection Department (1997-2015) and are shown in Fig. 1.1 Glass bottles make up the majority of the daily waste glass generation, which amounts to over 300 ton. Between 2012 and 2015, the government attempted to collect between 5-16% of this waste glass(Lu and Poon, 2019) . Recycling waste glass in concrete can serve as an alternative to natural aggregates and reduce potential risks to the hazardous environment. Although the use of recycled glass up to a definite percentage can be a better replacement for cement and aggregates.



Fig.1.1 Recovery percentage of waste glass in Hongkong (Lu and Poon, 2019)

EPA measures the generation, recycling, composting, combustion with energy recovery, and landfilling of glass materials. EPA used statistics from the Glass Packaging Institute's numbers on glass container shipment to estimate glass container generation. In 2018, glass generation in all products was 12.3 million tons in the United States, which was 4.2 percent of all MSW generation. The amount of recycled glass containers was 3.1 million tons in 2018, for a recycling rate of 31.3 percent. The total amount of combusted glass in 2018 was 1.6 million tons. This was 4.8 percent of all MSW combustion with energy recovery that year. In 2018, landfills received approximately 7.6 million tons of MSW glass.



Fig. 1.2: Glass Waste Management of U.S.A. (EPA-1960-2018)

In developing nations like Bangladesh, new high rise structures are built by razing old buildings, which were often made with masonry or reinforcing concrete structures. Bricks make up the majority of a historic building's construction. Because of this, a lot of recycled brick aggregate is being produced, which can be used in construction. An ancient government administrative building that was demolished to make room for a high-rise project is seen in Fig. 1.3. A lot of waste will be produced when a building like this is destroyed. This kind of waste cannot be broken down. Additionally, the production of new aggregates harms the ecosystem. Recycling is a potential solution for green concrete and sustainable building.



Fig. 1.3: Demolished of an old administrative building in Bangladesh

Reusing or recycling resources is a priority for developed nations. Although there are significant obstacles regarding strength, research into utilizing glass in concrete is still ongoing. Recycled concrete is also a well-established concept. Recycled concrete and glass are combined and thoroughly examined in this study. Along with the mechanical characteristics of concrete, the flexural characteristics of recycled glass brick concrete (RGBC) are also examined.

#### **1.2 Background and Motivation**

Reusing and recycling demolished concrete and construction materials has attracted immense interest to researchers and engineering communities because of their significant role in combating carbon emission, climate change and protecting natural resources. The global demand for concrete is rising to the point where natural resources are limited. The majority of existing buildings and structures in the subcontinent regions are made of masonry or brick. Many brick-oriented constructions in these regions will cease to function within the next 10 to 20 years, leaving no other choice than to demolish them. Additionally, the amount of waste glass is growing daily, and the majority of it is not biodegradable. In certain studies, discarded glass has been utilized in place of cement, fine aggregate, or even coarse aggregate. However, there is no research on the combined impact or behavior of recycled concrete glass (RCG) that may contribute to the commercial development of RBA-based glass concrete.

#### **1.3 Research Objectives**

The main objectives of the present research are

- i. To examine the physical properties of fresh and hardened concrete containing recycled brick as coarse aggregate (RBCA) and waste glass powder (WGP).
- ii. To investigate the mechanical properties of concrete containing RBCA and WGP as the replacement of coarse aggregate and fine content, respectively.
- iii. To investigate the flexural behavior of concrete beams comprising RBCA and WGP mixed concrete through four-point bending tests.

#### **1.4 Scope of the Thesis**

In this research work, recycled brick aggregate (RBA) and waste glass are core materials. Recycled Brick (RB) is collected from a 30 years old demolished building and then proceed a standard form to use as coarse aggregate in concrete. Another core materials of this research is Waste glass (WG) which is collected from Puran Dhaka waste tube light processing factory. As Waste Glass (WG) is used as the replacement of fine aggregate. So, the particle distribution curve is within the limit of ASTM C33 limits after several sieving. Besides a number of basic test will be occurred for deterimining the properties of aggregate. A proper mix design is prepared before casting procedure. For cyclindrical specimen there are fifteen batches and for prism beam and reinforced concrete (RC) beam contains twelve types of specimen .For determining workaibility of concrete slump and air content test is occurred according to ASTM C143 and ASTM C231 respectively. Besides compressive strength, splitting tensile strength and flexural test through center point loading is occurred according to ASTM C39, ASTMC496 and ASTM C293. ASTM C78 standard is followed for

four point flexural test of Reinforced Concrete (RC) beam. Linear Variable Differential Transducer (LVDT) is installed in the mid part and left mid part of the beam. Ultrasonic Pulse Velocity (UPV) of cylindrical specimen is occurred according ASTM C597. In addition Scanning Electron Microscopy (SEM) is occurred to get microstructure images of concrete.

#### 1.5 Organization of the Thesis

Apart from this chapter, the whole report is divided into five more chapters. The content of these five chapters is as follows:

Chapter 2 presents a review of nearly a hundred articles related to Recycled glass concrete.

Chapter 3 presents the methodology of targeted tests according to ASTM guidelines to meet the objectives of the research. It also presents the working procedure in time of casting and flexural test setup of beam. In addition, UPV and SEM test procedures are explored.

Chapter 4 presents the result of physical properties of aggregates, fresh and mechanical properties of recycled concrete containing glass (RCG). Targeted first two objectives are met in this chapter. Several code comparisons with the experimental value are also described. In addition, several comparison and result of UPV and SEM test procedure is explored.

Chapter 5 presents the result of the flexural test of the reinforced concrete (RC) beam. Several data from LVDT, Strain gauge, and UTM are presented. In addition, the ACI 318-14 code comparison with the experimental result is also explored.

Chapter 6 is the last chapter of this report which provides the concluding remarks. In the end, some recommendations are provided for future research purposes.

### . CHAPTER 2 LITERATURE REVIEW

#### **2.1 Introduction**

Consumption of natural resources like sand, brick, and stone should be kept to a limit for sustainability. Reusing or recycling can help ease the strain on natural resources due to this. Recycled aggregate is a usual or established research topic. Numerous investigators have attempted to replace coarse gravel, sand, and even cement with wasted glass. According to numerous research studies, employing glass increases strength up to a limit. However, another team of researchers demonstrated a contrary result on the use of glass on concrete. Fig.2.1 shows the statistics and percentage of research content used for review in this chapter. Near to a hundred articles have been reviewed in this chapter and most of the articles are from the last ten years. This chapter contains a review of types of waste glass, the chemical composition of glass, fresh properties of recycled glass concrete like workability, hardened properties of concrete, conductivity, ASR expansion, density, and properties of RBCA and at the end summary of this chapter.



Fig. 2.1 (a) Research statistics of literature on Recycled Glass Concrete (RGC), (b) researches reviewed in this article (2011- 2022)

#### 2.2 Waste Glass

The crushed glass exhibits an angular form, sharp edges, a smooth surface texture, and a greater aspect ratio than natural sand, as is seen in Fig.2.2 due to its brittle nature (Tan and Du, 2013). Here, in the Fig.2.2, different types and color glasses are shown. In addition, the micro structures of glass under the scanning electron microscopy (SEM) can be seen as shown in Fig.2.3 (Idir et al., 2020).



Fig. 2.2 Crumb Glass (a) appearance of natural and glass sand (b) flat glass





Fig. 2.3 SEM images of glass adapted with permission from Elsevier.

## 2.3 Chemical Composition of Glass

Chemical compounds of different types of glass, cement, and sand were represented in Table 2-1. It is evident that the main ingredient of all glass is SiO<sub>2</sub>. Interestingly, the presence of

 $SiO_2$  in types of all glasses is nearly consistent which is about 70% which is quite near to sand but much higher than that in cement. Besides, the major compound of cement is CaO (65-67) % whereas glass contains only 7-12 %. The significant presence of Na<sub>2</sub>O increases the density, refractive index, thermal expansion coefficient, bulk modulus, and Poisson's ratio of glasses, and decreases Young's modulus of the glass. On the other hand, K<sub>2</sub>O prevents the crystallization of glasses and promotes the final formation of glass.

Table 2-1: Chemical composition of cement, sand & different glasses (Du and Tan, 2014)(Idir et al., 2020)

Composition	Cement	Sand	Brown	Green	Clear	Flat	Hollow	Windshield
(%)			Glass	Glass	Glass	Glass	Glass	Glass
SiO <sub>2</sub>	20.80	88.54	72.08	71.22	72.14	72.16	69.89	70.11
Al <sub>2</sub> O <sub>3</sub>	4.60	1.21	2.19	1.63	1.56	0.68	1.92	1.86
Fe <sub>2</sub> O <sub>3</sub>	2.80	0.76	0.22	0.32	0.06	0.12	1.05	1.04
CaO	65.4	5.33	10.45	10.79	10.93	7.80	12.31	11.67
MgO	1.30	0.42	0.72	1.57	1.48	4.38	1.34	1.37
SO <sub>3</sub>	2.20	-	-	-	-	0.21	0.14	0.11
Na <sub>2</sub> O	0.31	0.33	13.71	13.12	13.04	0.21	0.16	0.07
K <sub>2</sub> O	0.44	0.31	0.16	0.64	0.62	14.46	13.18	13.76
TiO <sub>2</sub>		0.05	0.10	0.07	0.05			
Cr <sub>2</sub> O <sub>3</sub>			0.01	0.22				

It is evident from the ternary diagram of CaO-Al<sub>2</sub>O<sub>3</sub>-SiO<sub>2</sub> presented in Fig 2.4 that glass might be used in cement-based products. The pozzolonic reaction's primary reactive ingredient is  $SiO_2$  and with the addition of glass, the composition possesses its highest

amount (Harrison et al., 2020). However, the amount of CaO in cement is much greater than in glass powder which contributes major strength of mortar. Nowadays, mixes of Portland cement made from fly ash (FA) or granulated blast-furnace slag (GBFS) are extremely popular; and using waste glass in concrete may also be an option.



Fig.2.4 CaO-AL<sub>2</sub>O<sub>3</sub>-SiO<sub>2</sub> ternary diagram of glass powder, FA and GBFS, and Portland cement (adapted with permission from Elsevier) (Harrison et al., 2020)

#### 2.4 Fresh Properties of Glass Concrete

Fresh properties like workability, air content, and aggregate sizes are very important for the recycled glass concrete (RGC) to make it a feasible constituent of the concrete. Those properties decide the mixing procedure, internal bonding, water absorption, casting ability, etc.

#### 2.4.1 Workability

The ability of a concrete mixture to be worked is a well-known metric for assessing concrete. It has three characteristics, namely cohesion, water retention, and flow. Concrete's mobility is largely used to gauge how easily it can be worked. A typical slump test of concrete is presented in Fig. 2.5. Concrete workability can be separated into four categories based on the slump value: S1, tough concrete mixture i.e. very low workability (slump 10-40 mm), S2 medium workability (slump of 50 mm to 90 mm), S3 fluidity concrete (slump of 160 mm or less), and plastic concrete (slump greater than 160mm) (Jia et al., 2015).



Fig. 2.5 Typical slump test of concrete (Jia et al., 2015)

The slump value of waste glass concrete is presented in Table 2-2. The first column of the table indicates the amount of natural coarse aggregate (NCA) and the second column presents the coarse aggregate percentage replaced by recycled glass. Similarly, Fine Aggregate (sand) replaced by Recycled Glass is presented as (RGFA). A group of experimental results shows that incorporating glass as a replacement for fine aggregate provides better workability than normal one (Bostanci, 2019, Arivalagan and Sethuraman, 2021, Elavarasan, 2016, Haramkar et al., 2018, Jia et al., 2015, Ammash et al., 2009). In contrast, some researchers claimed a relatively lower slump value than that results in the control condition i.e., using sand (Arivalagan and Sethuraman, 2021, Pratheba.S, 2018, J.Premalatha 2019, Ibrahim, 2020). This can be described by the fact that the fine glass powder absorbs a larger amount of water compared to that in the normal concrete and hence reduce the slump value. However, the variation of workability for both types of cases is negligible. Overall, it can be concluded that the application of Recycled Glass Coarse Aggregate (RGCA) yields a comparatively lower slump value (Gerges et al., 2018).

The addition of glass sand did not modify or trend the slump readings significantly. Although the slump of concrete would be lessened by the sharper edge and more angular form of crushed glass sand (Limbachiya, 2009, Du and Tan, 2014, Park et al., 2004, Taha and Nounu, 2009). The cement paste's workability was slightly affected by the RWG composition since the glass particles wouldn't absorb water from it. Researchers followed the European standard EN 12350-2 is followed in conducting the consistency test. Fig. 2.6 (a) shows the slump values of concrete containing different percentage (%) of recycled glass for different

concrete grades. The result indicated that with the increase of glass content in C30 concrete, the slump value initially decreased (up to 50%) and increased for further addition of glass. The behavior is opposite for C45 and C60 concrete. In Fig. 2.6 (b) first three mixtures are showing values that are in the acceptable boundaries (8-12cm). The exception is the result from mixture 4 which may happen due to the presence of the super-plasticizers (Grujoska et al., 2020). The results are showing that the slump is decreasing with the increase of the glass in the mixture. This phenomenon can be described by the fact that the glass particles are absorbing comparatively lesser amounts of water than natural aggregate particles(Grujoska et al., 2020).

Coarse Aggregate		Fine Ag	gregate	Slump	Ref.
NCA (%)	NCA (%) RGCA		RGFA (%)	(mm)	(Gerges et
	(%)				al., 2018)
100,67,50,33,0,	0,33,50,67	100,100,100,	0,0,0,0,0,	100,100,70,7	(Bostanci,
100	,100,0	100, 100,0	100	0, 30,90	2019)
100	0	100,90	0,10	68,75	(Ibrahim,
					2020)
100	0	100,95,90,85	0,5,10,15,2	100,98,96,	(Elavarasan,
		, 80,75,	0, 25,30,	95,94,92,	2016)
		70,65,60,55,	35,40,45,5	20 27 26 25	
		50	0	09,07,00,0 <i>3</i> , 85	
				0.5	
100	0	100,90,80,70	0,10,20,30	65,68,73,60	(J.Premalath
					a 2019)
100	0	100,90,80,70	0,10,20,30	52,48,60,59	(Arivalagan
					and
					Sethuraman,
					2021)
100	0	100,90,80,70	0,10,20,30	35,53,65,72	(Pratheba.S,
					2018)

Table 2-2. Slump values of concrete containing glass concrete

100	0	100,90,80,70	0,10,20,30	52,38,43,48	(Haramkar et
					al., 2018)
100	0	100,90,80,70	0,10,20,30	35,53,65,72	(Ammash et
					al., 2009)
100	0	100,90,80,70	0,10,20,30	25,29,34,40	(Arivalagan
					and
					Sethuraman,
					2021)
100	0	100,90,80,70	0,10,20,30	97,95,72,65	(Haramkar et
					al., 2018)
100	0	100,90,80,70	0,10,20,30	35,53,65,72	(Jia et al.,
					2015)
100	0	100,50,0	0,50,100	47.8,48.2,54.	(Haramkar et
				6	al., 2018)





The concrete flow test is another measure of workability that usually determines the possibility of a rise in the mortar base perimeter following 15 drops on a flow table. As can be seen from Fig. 2.7, the flow ability of glass beach mortar decreased as the glass content increased. Flow rates for full-bodied brown, green, transparent, and mixed-color glass beach

mortar, were independently 75%, 90%, 60%, and 82%, respectively of those for typical beach mortar (Tan and Du, 2013). This phenomenon of RGC can be correlated to the slump value where it is observed that the presence of glass powder reduces the slum value and hence flow of concrete also decreases.



Fig.2.7 Flow of cement mortar containing different proportions and types of glasses (Tan and Du, 2013).

#### 2.4.2 Air Content

The percentage of glass sand in concrete influences the air content. It is observed from Fig. 2.8(a) that the air contents vary in a range of 0.75% - 1.5% for all concrete grades. With a glass content of 25%, the air content is observed to be reduced, though the value increases linearly with higher glass contents (Du and Tan, 2014). The behavior is very much closed to that observed by Park et al. (Park et al., 2004) where they claimed that the air content of glass sand concrete rises as high as 41% for the glass content of 70%. In contrast, the air content of the glass sand mortar is nearly constant (air content of 3%) up to a glass content of 75% for different glasses as presented in Fig. 2.8(b) though the value rises sharply for the glass content of 100% (complete replacement of fine content), particularly of the green and clear glass. The higher air content was observed for very high glass content because the irregular and sharper edges of glass particle cause voids inside the mix which allows more air retention at the glass particles' surface (Tan and Du, 2013).



Fig. 2.8 Air content with glass sand content for different (a) concrete grades and (b) colors of glasses.

#### 2.5 Hardened Properties of Concrete

#### 2.5.1. Compressive Strength

Compressive strength is the major concern of concrete since its structural applications directly depend on the compressive strength properties of concrete. Numerous 28 days compressive strength test data were collected from various peer-reviewed publications as shown in Table 2-3. Here, recycled waste (RG) is used as a replacement for Cement, Fine aggregate, and Coarse aggregate. Using RG as a replacement of natural coarse aggregate (NCA) is termed as recycled glass concrete aggregate (RGCA) results in a lower strength compared to that obtained from normal fresh concrete. Usually, RGCA replacement up to 30% yields a significant change of strength which is about 8-12% (Nwofor and Ukpaka, 2017, Gerges et al., 2018, Sopov et al., 2020). Srivastava et al (2014) revealed that RGCA compressive strength was increased with the increasing addition of coarse aggregate up to replacement of 20%. However, their claim was found only for a lower percentage of NCA replacement by glass.

Many researchers used recycled glass's finest version as a replacement for cement. Up to 10% cement replacement yields a reasonable improvement in concrete inters of strength parameters in a range of 6 to 25% (Abbas et al., 2021, Ling and Poon, 2011). The increase of strength happens because the relatively higher fineness of glass powder fills the gaps among other constituents of concrete at nanoscale that ultimately improves the compressive

strength of concrete. On the opposite hand, different types of w/c ratios and various % of glass powder showed a rapid change in compressive strength (Salih and Alrubaie, 2018). Many researchers showed much interests in employing RG as a replacement for the fine aggregate (RGFA) of concrete. Most of them revealed that using 20-30% RG as a replacement for fine aggregate (mainly sand) in concrete yields a significant improvement in compressive strength (Jeba Samuel, 2018, Arivalagan and Sethuraman, 2021, Elavarasan, 2016, Pratheba.S, 2018, Rjoub and Tamimi, 2019, Alducin-Ochoa et al., 2021, J.Premalatha 2019, Ibrahim, 2020, Gautam et al., 2012). In addition, they also claimed that there is no significant reduction in compressive strength due to the presence of RGA as a fine content of concrete. However, if the replacement level exceeds 30%, the strength properties may decline as observed in many research works. This phenomenon may be described by the fact that at a relatively lower percentage fine glass may fill the gap among the aggregate and increase the overall compressive strength. On the other hand, higher glass content in concrete yields smaller bonding between the fine content because of the smoother surface of the fine glass, and hence lower compressive strength can be obtained.

	RGCA	RGFA	Compressive Strength	Ref.
W/C ratio	(%)	(%)	(MPa)	
0.55	0,33,50,67,100,	0,0,0,0,0,100	25.2,22.7,18.9,18.3,10.	(Gerges et
	0		8,20.9	al., 2018)
-		0,10,25	32,36.8,37.9	(Alducin-
				Ochoa et
				al., 2021)
0.53,0.52,0.50,	-	0,5,10,15,20,	37.80,31.30,32.55,33.3	(Salih and
0.48,0.50,0.54,0.56		25, 30 (Wt. of	5,32.10,29.70,28.40	Alrubaie,
		Cement)		2018)
0.45,0.45,0.64,0.64	-	0,10,0,10	48,42,36,35	(Bostanci,
				2019)
0.50		0,5,10,15,20,	27,29,31,27.5,20,16,14	(Aper
		30,40		ZAVA,
				2020)
		0,10,20,30,	30.33,31.33,31,29.66,	
0.50		40,50	29.33,24.33	(Gautam et
				al., 2012)

Table 2-3: Influence of recycled glass on Compressive Strength of glass concrete

0.50	-	0,5,10,15,20,	34.2,36.5,37.2,37.7,38.	(Ibrahim,
		25,30,35,40,45,5	12,38.87,34.06,32.72,3	2020)
		0	2.53,31.12,28.09	
0.50	-	0,100	51.2,34.7	(Czapik et
				al., 2021)
0.41	-	0,25	40,46	(Rjoub and
				Tamimi,
				2019)
0.50	-	0,10,20,30	30.33,31.33,31,29.66	(Elavarasa
				n, 2016)
		0,10,20,30	26,27.10,27.50,28.80	(J.Premalat
				ha 2019)
		0,10,20,30	25,36.80,25.80,24	(Pratheba.S
				, 2018)
0.50	0,0,30,30	0,30,30,0	30,27,30,43	(Sopov et
0.50	0.10.00.00.10			al., 2020)
0.50	0,10,20,30,40,		29.67,31,29.67,25.33,2	(Srivastava
0.45	50	0.10.00.00	7.33	et al., 2014)
0.45	-	0,10,20,30	27.3,29.2,30.2,29.8	(Jeba
				Samuel,
0.40		0 5 10 15	22 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	$\frac{2018}{(Lakash at)}$
0.40	-	0,5,10,15	22.3,23.2,22.3,21.38	(Lokesn et al. 2017)
0.45		0 10 20 20	22 57 26 99 27 11 22 7	$\frac{\text{al., 2017}}{(\Delta rivelege)}$
0.45	-	0,10,20,50	6	(AIIValaga n 2021)
0.30	0 25 50 75 100		36 91 40 64 43 49 44 3	(Jabal et
0.50	0,25,50,75,100		5 43 10	(Jabar Ct)
0.45-0.5		0 10 15 (Wt of	43 54 52	(Ubeid et
0.10 0.0		Cement)	10,01,02	(00001a et al., 2020)
0.55-0.6	_	0.10.20.30	29.68.33.03.27.14.25.1	(Haramkar
		-,,	1	et al., 2018)
0.45	_	0,10,15	21.45,19.22,19.1	(Mishra et
		, ,	, ,	al., 2020)
0.50	-	0,20,40,60	33.45,32.49,30.9,30.71	(Nwofor
				and
				Ukpaka,
				2017)
0.50	0,20,40,60		33.45,34.46,31.93,26.7	(Nwofor
			4	and
				Ukpaka,
				2017)
	-	0,2.5,5,7.5, 10	36.8,39.5,41.5,40.8,39.	(Abbas et
		(Wt. of Cement)	2	al., 2021)

0,10,30,50,100	46.19,39.95,31.17,14.1 6 4 23	(Drzymała et al. 2020)
 0,50,100	47.8,48.2,54.6	(Wang et al., 2014)

Based on previous research, the compressive strength of RGC specimens is presented in Fig. 2.9 and 2.10 where the waste glass was employed as a replacement for fine aggregate and cement, respectively. In Fig .2.9 (a) Samples ID I-V were prepared based on the RG replacement of coarse aggregate with different proportions of 0%,33%,50%,67%, and 100%, respectively, and specimen VI was fine glass sand. It is observed that there is a clear decline of compressive strength (10%-58%) with the increase of RGCA percentage (33%-100%). Replacing 100% glass as a fine aggregate in specimen VI reduced the compressive strength by 17%.

Previously, it was evident that 10% RGFA provides improved strength than the normal concrete which was 19- 25% approximately as shown in Fig.2.9 (b) and (c). Besides, after using 30% of RGFA compressive strength began to decline which is similarly evident in the references However, there are pieces of evidence where it is also found that with the rise in the proportion of glass particles, there is a decline in compressive strength regardless of the color of the glass at 7 days ages of hardened concrete as presented in Fig. 2.10 (d). In addition, WG levels of 15, 30, and 60% reduced concrete's 28 days compressive strength by 2%, 8%, and 13%, respectively, compared to the control concrete as shown in Fig. 2.9 (e).

It can be observed from the Fig. 2.9 (f) that as the WG particle size is decreasing, the strength properties are increasing. This may happen due to the fact the finer glass particles can travel finest pores of the aggregates and hence make the concrete more compact. The time dependent strength increment is quite significant for fine glass powder. The outcome of this research was further proved by where they claimed that with the addition of glass powder, an excellent pozzolanic effect is always seen, with a 180-day compressive strength that is at least 80% higher than the associated 28-day compressive strength. This may happen because the increased amount of CaO in this combination (high glass content) continues the reactions for a longer period which ultimately increases the strength at 180 days.

Quite a few researchers have used waste glass as a replacement for cement in concrete. Glass contains almost 72% SiO2 and is near about 12% CaO. For CaO and SiO2 content, glass has adhered binding properties compared to that present in the sand only. The compressive strength at different mixing ratios is presented in Figure 2.10 (a-f) (Limbachiya, 2009, Bentchikou et

al., 2017, Hasani et al., 2022, Du and Tan, 2014, Ling et al., 2011, Aper ZAVA, 2020, Herki et al., 2017, Ling and Poon, 2011, J.Premalatha 2019, Gerges et al., 2018, Corinaldesi et al., 2016, Tan and Du, 2013, Rahim et al., 2014). The results acknowledged that the compressive strength increased proportionally with the curing process because the hydration process was still continuous until getting the full strength of concrete. However, in the case of employing a mixture of waste glass as a partial replacement of cement, seemed to be a reduction in compressive strength as shown in Fig. 2.10 (a-c,d).





Fig. 2.9 Influence of Glass as a replacement of sand on the Compressive Strength of RGC for different (a,e) mixtures (b, c) ages (d) colors (f) glass powder sizes

Fig. 2.10 (a) shows a 50-60% reduction of mortar strength for 11-15% glass mixed mortar. Fig. 2.10 (b) and (c) revealed that the reduction of mortar strength is nearly 20-30% for the replacement of 20% glass regardless of the glass colors. In comparison to other concrete, 30% RG concrete demonstrated a lesser strength gain. It is also evident that the strength can be improved with the addition of fly ash. This rationale can be explained as the reduction in lime after the reaction that played a serious role in the production of calcium silicate hydration which is liable for the strengthening of concrete. Also, the presence of impurities in each of the crashed waste glasses affected the strength properties of the concrete. However, there is evidence that the incorporation of WG as a replacement for cement can improve the mortar strength, particularly at the 20% replacement as shown in Fig. 2.10 (d). Another research outcome presented in Fig. 2.10 (e) and (f) claimed that the 20% replacement with a ratio 0.26% W/C ratio shows the maximum compressive strength for both 56 and 112 days including a variation with it different types of w/c ratios at 56 days and 112 days in Fig. 2.10 (e, f). Here, compressive strength is rising as compared to normal concrete for up to 20% glass powder before beginning to decline.



Fig. 2.10. Influence of varying waste glass proportions on the Compressive Strength of cement mortar for different (a, b, d) ages (c) glass color (e, f) W/C ratio (khalil et al., 2018, Schwarz et al., 2008, Wilson et al., 2019, Özkan and Yüksel, 2008, Al-Zubaidi et al., 2017, Li et al., 2021, Tamanna and Tuladhar, 2020)

#### 2.5.2 Splitting Tensile Strength

Quite a few studies have been conducted on the split tensile tests to determine the tensile properties of concrete made from recycled glass concrete (RGC) as presented in Table 2-5. The table presents the split tensile strength of RGC where RG was used either as the replacement of fine content of the coarse aggregate. Here, concrete with RG as fine aggregate (RGFA) shows that the tensile strength is comparatively smaller than that of normal concrete (Arivalagan and Sethuraman, 2021, Rjoub and Tamimi, 2019, Ammash et al., 2009). In contrast, some research showed that increasing the value of tensile strength with the addition of RGFA (Elavarasan, 2016, Pratheba.S, 2018, Lokesh et al., 2017). Besides, it is also evident that the tensile strength increases with the addition of RGCA up to a certain percentage like 20-30%, for cement and the outcome revealed that the compressive strength can be increased by as and decreases with the higher RGCA (Jabal et al., 2021, Ibrahim, 2020). Recycled glass is additionally used as a replacement high as 15% of replacements than that of the control specimen. Practically, the concrete strength is very much sensitive to the size, shape, glass type, and surface roughness of the recycled glass.

W/C	RGCA%	RGFA%	Tensile Strength (MPa)	Ref.
0.5	0,5,10,15,20		2.98,3.27,3.29,3.32,3.36,3	(Ibrahim, 2020)
	,25,30,35,		.43,2.95,2.86,2.66,2.38,2.	
	40,45,50		35	
0.41		0,25	4.1,3.9	(Rjoub and Tamimi,
				2019)
0.5		0,10,20,30	2.3,2.52,3.21,3.1	(Elavarasan, 2016)
		0,10,20,30	3.45,3.40,3.46,3.04	(Arivalagan and
				Sethuraman, 2021)
		0,10,20,30	2.9,3.83,3.8,3.24	(Pratheba.S, 2018)
0.4		0,5,10,15	2.9,3.1,2.74,1.9	(Lokesh et al., 2017)
0.45		0,10,20,30	3.45,3.40,3.46,3.04	(Arivalagan and
				Sethuraman, 2021)
0.3	0,25,50,75,1		3.12,3.40,4.62,4.87,4.90	(Jabal et al., 2021)
	00			

Table 2-5: Split Tensile Strength of Concrete containing RG as Coarse or Fine Aggregate

0.45-		0,10,15	3.9,4.8,5	(Ubeid et al., 2020)
0.5		(Cement)		
	0,10,30,50,1		8.3,6.9,7.43,4.55,3.84	(Drzymała et al., 2020)
	00			
0.5		0,10,20,30,	3.75,3.6,3.55,3.48,1.9	(Ammash et al., 2009)
		40		

The tensile strength of RGC composed from different publications suggested that primarily, the split tensile strength decreases with the increase of RGFA as presented in Fig. 2.11(a) and (b). This may happen because the smoother surface of glass creates relatively weaker bonding between waste glass and other constituents of concrete that directly effects adversely on the tensile strength of the RGC. However, some researchers claimed that RGFA up to a certain percentage, improves the overall tensile strength of RGC regardless of water cement ratio (W/C) and age of the concrete. The maximum value of split tensile strength is observed at 20% fine replacement by WG at W/C ratio of 0.26. Fig. 2.13 (e) shows that the tensile strength decreased after 50% WG replacement whereas it was observed that the 100% replacement yields the maximum tensile strength for all concrete grades.



![](_page_35_Figure_0.jpeg)

Fig. 2.11. Split tensile strength of RGC for different mixing ratios (a, b) with age (c) W/C ratio and days and (d) concrete grade (Du and Tan, 2014, J.Premalatha 2019, Kalakada and Doh, 2020, Al-Zubaidi et al., 2017, Li et al., 2021, Tan and Du, 2013, Grujoska et al., 2020)

#### 2.5.3 Flexural Strength

Bending strength tests are necessary to understand the flexural properties of concrete. Limited but very useful flexure test data were collected from the literature as presented in Table 2-6. Very few evidences are found in the literature where RG has been used as a replacement for coarse aggregate. They claimed that the flexural strength increases as the RGCA increases in the concrete. On the other hand, RGFA up to 20% yields an improvement in flexural strength (Arivalagan and Sethuraman, 2021, Sunarsih et al., 2021). However, one more research showed that flexural strength increased up to 5% RGFA only (Lokesh et al., 2017). Some researchers used recycled glass particles as a replacement for coarse aggregate, and the results revealed that flexural strength increased as the amount of recycled glass increased. (Jabal et al., 2021). This phenomenon can be described by the mechanism that for a smaller percentage of glass the weaker bonding is not dominant as the glass powder fills the fine pores of other aggregates and hence the flexural strength increases with the addition of RG.
W/C	RGCA%	RGFA%	Flexural Strength	Ref.
			(MPa)	
0.5		0,100	8.3,5.1	(Czapik et al.,
				2021)
0.4		0,5,10,15	5.5,6,4.7,4.21	(Lokesh et al.,
				2017)
0.45		0,10,20,30	4,4.01,4.6,3.9	(Arivalagan and
				Sethuraman,
				2021)
0.3	0,25,50,75,100		4.06,5.63,6.92,8.78,	(Jabal et al.,
			9.21	2021)
0.45-0.5		0,10,15	4.41,5.18,4.83	(Ubeid et al.,
		(cement)		2020)
		0,2.5,5,7.5,10	4.96,5.11,5.21,5.34,	(Abbas et al.,
		(cement)	5.13	2021)
		0,10,20,30,40	2.06,2.41,2.94,2.74,	(Sunarsih et al.,
			2.45	2021)

Table 2-6: Flexural properties of glass concrete

A good number of flexural strengths test data are collected and a part of them are presented in the Fig. 2.12 (a-d). Figure 14a shows that the flexural strength of concrete reduces with the increase of glass percentage though the strength can be improved with the addition of admixtures for the case of 50% glass. Another research by (Du and Tan, 2014) claimed that with the addition of RG, flexural strength improves even up to 100% regardless of concrete grade as shown in Fig. 2.12 (b). Fig. 2.12 (c) and (d) show the influence of RG% on the flexural strength for different W/C ratios and ages. As presented in the figures, the experimental investigation claimed that the maximum flexural strength of RGC is found for 25% RG as a replacement of fine content for all W/C ratios and also at both 56 days and 112 days. The result can be correlated to the compressive strength and tensile strength test data where it is also observed that at a low percentage of RG the strength improves but at a higher percentage, it usually decreases as the glass percentage increases. The reason behind the improvement at a low percentage is that the fineness of the glass powder makes more integrated concrete.



Fig. 2.12. Flexural strength of RGC with (a) varying sand replacement by RG, (b) with different concrete grades (c, d) with age and W/C ratios (Bentchikou et al., 2017, Du and Tan, 2014, Ling et al., 2011, Ling and Poon, 2011, J.Premalatha 2019, Grujoska et al., 2020)

## 2.6 Conductivity

The conductivity of glass concrete has been studied quite a few times by several researchers as presented in Fig. 2.13 (a-b). Previously conducted rapid chloride penetration tests (RCPT) have been highlighted in Fig. 2.13 (a, c), and here Dnnsm was a non-steady state migration coefficient. Except for the low glass content in C30 and C45 concrete as shown in Figure 16(a), the entire charge passed at 28 days for an equivalent w/c. The value was found significantly reduced by increasing the glass sand content. For C30 and C45 concrete mixtures, plain or 25% glass concrete exhibited a charge above 2000 coulombs, indicative

of moderate permeability consistent with ASTM C1202. The quantity of transferred charge is decreasing as the number of days and % of glass content increase, which implies permeability decreases, as presented in Fig. 2.13 (b, d). Fig. 2.13 (c) displays further RCPT data. All of the mortar specimens had high total charge passing values, with the typical sand mortar displaying the greatest value of 6764 ( $\pm$  966) Coulombs. This is due to the porous nature of cement paste and the absence of coarse particles. The charges % that were migrated through the 50% brown, green, clear, and mixed glass sand mortar compared to the conventional sand mortar was found 93%, 69%, 71%, and 64%, respectively. According to a prior research by Kou and Poon (2009), conductivity values were found comparatively lesser value, where they reported a 60% reduced charges can be migrated when 45% of the natural sand is substituted with glass sand (Kou and Poon, 2009).



Fig. 2.13 Conductivity of glass concrete for (a) concrete grade (b) RG (c) different colors (d) different ages

### 2.7 ASR expansion

Alkali-Silica Reaction (ASR) expansion behavior is shown in Fig. 2.14. In fig. 2.14 (a), all concrete mixtures showed expansion but 0.1%, and 0.2% at 14 and 28 days, respectively. Here at 49 days, only the reference specimen crossed 0.2% expansion. An identical trend was also observed in Fig. 2.14 (c) (Schwarz et al., 2008). It is claimed by quite a few researchers that the presence of glass in concrete is innocuous in alkali reactivity (Limbachiya, 2009, Zhu et al., 2008, Du and Tan, 2013). Fig. 2.14 (b) also shows similar evidence as Fig. 2.14 (a) where it is evident that the ASR expansion is very much dominant after the age of 15 days. . Here, the ASR expansion grew significantly as the RG content rose from 0% to 100%. Even though 10% MK had previously been used to replace cement at the age of 14 days, all of the mortar bar mixtures had ASR expansion within the permitted limits (0.10%), except for the M10R100 mix. It is also observed that ASR expansion grew significantly as the RG content rose from 0% to 100%. Content rose from 0% to 100% (Ling et al., 2011).

Research outcomes on the expansion of the mortar bar specimen, containing 100% WG, nearly 0.2% before 3 days have been displayed in Fig. 2.15. Expansions were just 0.2% after 4 days when the WG content was 0%. As a result, no significant alkali-silica reactions were observed when WG content decreases for this reason (Baykal, 2003).





Fig.2.14 (a-d) Expansion behavior of different RGFA containing concrete with time Figure 20 (a-c) are adapted with permission from Elsevier (Du and Tan, 2014, Ling et al., 2011, Schwarz et al., 2008, Liu et al., 2015)



Fig. 2.15. Expansion behavior of different % of RG as a replacement of Coarse Aggregate

# 2.8 Density of RG concrete

Density of concrete is a very important parameter for its practical application. Fig. 2.16 (ad) and 2.17, present the density of glass concrete mixed as a replacement of fine and coarse aggregate, respectively. Different types of grades or different types of RGFA showed a lower density compared The to that of the normal concrete as presented in Fig. 2.16 (a, b, d). The figures also show that the density of RGC decreases as the percentage of WG increases in the concrete. Moreover, the dry density of RGC is smaller than that of the fresh condition. However, a lower particle size showed better density compared to a bigger one as suggested by (Kalakada and Doh, 2020). Since glass sand has a lower density than fresh sand due to its lower specific gravity the hardened density of RGC is also lower than that of normal concrete (Du and Tan, 2014).



Fig. 2.16 (a-d) Fresh and hardened density of concrete with WG for (a) concrete grade (b) time and proportions (c, d) mixtures (Du and Tan, 2014, Herki et al., 2017, Kalakada and Doh, 2020, Tan and Du, 2013, Rahim et al., 2014)

Fig. 2.17 demonstrated the densities of concrete made from gravel and RGCA. The cubic samples of glass concrete exhibited approaching gravel concrete after 28 days due to their sparkling look and smoothness. Except for 20% glass replacement, all of the samples of concrete displayed denser specimens, which may be attributed to the proper placement of glass as an aggregate in the cement-sand combination (Fahad and Ali, 2020).



Fig. 2.17: Density of Concrete containing RG as replacement of CA

## 2.9 Properties of Recycled Brick Aggregate (RBA)

Several studies are available where normal brick aggregate (as coarse and fine) properties were examined and even compared with stone chips/aggregates. However, research on recycled brick aggregates is still limited. One of the major issues with recycled brick aggregates is the high water absorption capacity. Adhered old mortar paste in recycled aggregates tend to absorb water from the concrete mix and leaves less water for binder reaction (Gerges et al., 2018). This high water absorption together with the irregular shape of recycled aggregates may cause lower slump/workability of fresh concrete (Tittarelli et al., 2018). It is also worth mentioning that most old mortar paste structure are porous, i.e., the presence of lots of micro voids, which is also a reason for the high water absorption of this type of aggregates (Tan and Du, 2013).

However, a contradiction in the mechanical properties of concrete with brick aggregate is reported by several authors. Bangwar et al. found about 10% lower compressive strength of concrete at 28 days of testing when 50% of the coarse aggregates was replaced by brick aggregate. Similarly, Debieb & Kenai reported about a 5 to 10% lower strength for 25% fine brick aggregate replacement. Authors have found a maximum 30% strength reduction for full fine aggregate replacement. Conversely, Another research showed that the utilization of fine crushed brick aggregate in concrete up to 25% with natural sand showed the same 90-day strength as the control mix without any brick aggregate (Agwa et al., 2020). It was further proved by Mohammed et al. that brick aggregates do improve the mechanical properties of

concrete. The compressive strength and Young's modulus of concrete made with brick aggregate were found to be higher than the concrete made with stone aggregate. The authors also concluded that recycled brick aggregates show better performance than the normal brick aggregates though the abrasion and absorption capacity of recycled brick is higher. Although the durability of brick aggregate concrete is not covered in this study, however, some previous studies mentioned that the durability of concrete may be susceptible to brick aggregates.

### 2.10 Summary

A good number of researches have already been conducted on the waste glass as an aggregate of concrete. However, there are some key issues that are need to be addressed before its practical application. The fresh and hardened properties of concrete with the combination effect of recycled brick aggregate and RG were not analyzed before. Non-destructive test like Ultrasonic Pulse velocity in such concrete was not evaluated previously. The application of RBA-GC in structural elements such as beams was not analyzed previously. Based on the comprehensive exploration of the existing literature, the key points can be summarized as follows

- a. Using RG in concrete as a cement or coarse aggregate provides comparatively better workability than the normal one. Though the workability of concrete containing RG as fine replacement decreases with the increasing percentage of RG, it improves workability up to a certain percentage. Overall, the performance depends on the shape, size, and type of RG used in concrete.
- b. Incorporating RG up to 20-30% in concrete as a fine aggregate gives a positive value of compressive strength compared to a normal one. However, it shows decreasing trends when it is used as a replacement for cement or coarse aggregate. The strength properties of RG mixed concrete vary with the different types of glass, particle sizes, and shapes.
- c. Due to lower specific gravity, RGFA showed lower density compared to sand in concrete though the finer particle size of RGFA contributes to a better density and hence relatively lighter concrete. The case of RGCA showed a relatively denser specimen which is close to the normal one.

### CHAPTER 3 METHODOLOGY

### **3.1 Introduction**

This chapter includes the overall methods and techniques used for conducting the experimental investigations. Firstly, the physical properties of the waste glass, recycled brick aggregate and other natural aggregates are determined through the ASTM standards. The methods involved in the specimen preparations for physical properties are also highlighted. Test plans and procedures for the mechanical properties of aggregate, mixture proportions, including test matrix, fresh properties of concrete, casting, curing and notation process are presented in this chapter. Specimen preparation and test methods for compressive strength, split tensile strength and flexural properties of recycled brick aggregate concrete with recycled waste glass are also discussed. In addition, test procedure and protocol for scanning electron microscopy (SEM) has also been conducted. Finally, test setup, data acquisition technique and specimen preparation for the four-point bending test of reinforced concrete beam specimens are also demonstrated in this chapter.

#### 3.2 Collection Procedure and Physical Properties test of Core Materials

An experimental program has been designed following the research objectives and scope of work. Before going to determine the mechanical properties, physical aspects of aggregate are required to be identified, especially for conducting a proper mix design. In this part, a series of basic tests for aggregated such as Specific Gravity (S.G.), Fineness Modulus (F.M.), unit weight, etc. test are demonstrated to understand the physical properties of aggregates. In this research work brick aggregate and waste glass are core materials which are shown in Fig.3.1 Coarse aggregate (CA) is partially replaced by the recycled brick aggregate generated from the demolished concrete of a 30-year-old first-generation building as shown in Fig. 3.1 (a). After eliminating the debris, recycled CA was obtained through the process of screening, sifting, and washing of the demolished concrete. Normal brick coarse aggregate (NBCA) is obtained from the locally available fresh 1st class brick crushed through the manual process. Both recycled CA aggregates and normal brick aggregate (NBA) were sieved to make them 4.75 to 19mm downgrade following ASTM C136-14. The table shows the physical properties of aggregate like sand, brick, recycled brick, and glass.

In this investigation, waste glass is used as a partial replacement for fine aggregate. The locally available well-graded natural sand with a maximum grain size of 4.75 mm satisfies the need for natural fine aggregate. Sand is graded and its sieve analysis is carried out in accordance with ASTM C136-14. Waste glass powder (WGP) was collected from the old town where wastes are processed on demand. Basically, here waste glass is extracted from tube glass and then processed into powder form to make sure it fit to be used as the fine aggregate as shown in Fig. 3.1. In this study, waste glass is used in powder form as a partial replacement of sand. Fig.3.2 represents the particle distribution curve of sand and waste glass powder (WGP) which is within the recommended standard of ASTM C33.



Fig. 3.1: Core materials of this research (Recycled Brick and Waste Glass)

Table 3-1: Physical properties of aggregate

Variables	NBCA	RBCA	Sand	RG
Specific Gravity	2.1	2.1	2.59	2.46
Absorption	19.21	18.23	4	3.8
Capacity (%)				
Fineness	6.6	6.6	2.6	2.56
Modulus (F.M.)				
Unit Weight	1305.02	1310.54	1646.44	1521
$(kg/m^3)$				

Several trials were made to get the particle distribution according to ASTM standard. In this research recycled glass is used as a replacement for sand. there, the particle distribution curve of glass should be within the limit and quite similar to sand. Fig. 3.2 shows the particle distribution curve of recycled glass and sand which is within the limit of ASTM standards.



Fig 3.2: Particle Size Distribution Curve of WG and Natural Sand

## **3.3 Mixture proportions**

Two types of specimen are planned for determining the compressive strength properties of recycled brick aggregate based concrete with glass. At first, fifteen different concrete mixes are prepared for cylindrical specimen to support comparative analysis. Then based on this result

another twelve batches of prism beam (100mm×100mm×500mm) has been cast. A total of twelve batches mixture were prepared for determining the flexural strength of the prism beam. A target compressive strength of 25–30 MPa (i.e., 28 days concrete strength) is commonly used in designing reinforced brick concrete structures in Bangladesh depending on the category of construction. For this reason, the designed compressive strength was proposed to be 25 MPa to provide more realistic and comparable evidence to the concrete industry. The total Portland cement content and the effective water-cement ratio were kept constant at 340 kg/m<sup>3</sup> and 0.48, respectively. As a result, the variations are in the percentage of natural coarse aggregate, RCA, sand, WG, and cement. The coarse and fine aggregates used in the concrete are in saturated surface dry (SSD) condition.

The percentage of RCA replacement is 10% and 30% which are replaced on a weight basis. In the previous study, it was shown that RCA replacement up to 30% had no large effect on concrete strength, but thereafter there was a gradual reduction in strength as the RCA content increased [4,8,9,36]. Some researchers have mentioned not to exceed 30% replacement level to maintain standard requirements for 5% absorption capacity of aggregates [49]. This is why the RCA replacement level is considered as 10% and 30%. Fine aggregate is replaced from 10 to 100% by waste glass (WG). Fourteen batch details are shown in Table 3-3. For example, RC30G20 means here recycled brick aggregate has replaced about 20% of total coarse aggregate and fine aggregate has replaced about 20% by waste glass. From batch no.4 to 15 amount of recycled concrete (RC) has been kept constant and the amount of glass has been changed to see the proper effect of waste glass powder on concrete.

Batch identification name: RC <sub>x</sub> G <sub>y</sub>	Description
RC	Recycled Concrete
X	Amount of Recycled concrete (Coarse %)
G	Glass
У	Amount of waste glass (Fine %)

Table 3-2: Description of specimen identification

Batch no.	Batch code	NCA (kg)	RCA(kg)	Sand(kg)	WG(kg)
1	$RC_0G_0$	860	0	690	0
2	$RC_{10}G_0$	774	86	690	0
3	$RC_{20}G_0$	688	172	690	0
4	$RC_{30}G_{0}$	602	258	690	0
5	$RC_{30}G_{10}$	602	258	621	69
6	$RC_{30}G_{20}$	602	258	552	138
7	RC <sub>30</sub> G <sub>25</sub>	602	258	517.5	172.5
8	RC30G30	602	258	483	207
9	RC30G40	602	258	414	276
10	RC30G50	602	258	345	345
11	RC30G60	602	258	276	414
12	RC30G70	602	258	207	483
13	RC30G80	602	258	138	552
14	RC30G90	602	258	69	621
15	RC30G100	602	258	0	690

Table 3-3: The proportions of aggregates for concrete mixture per cubic meter (Cylindrical)

Table 3-4: The proportions	of aggregates for	or concrete mixture	per cubic meter	(beam)
----------------------------	-------------------	---------------------	-----------------	--------

Batch no.	Batch code	NCA (kg)	RCA(kg)	Sand(kg)	WG(kg)
1	RC30G0	602	258	690	0
2	RC <sub>30</sub> G <sub>10</sub>	602	258	621	69
3	RC30G20	602	258	552	138
4	RC <sub>30</sub> G <sub>25</sub>	602	258	517.5	172.5
5	RC30G30	602	258	483	207
6	RC <sub>30</sub> G <sub>40</sub>	602	258	414	276
7	RC <sub>30</sub> G <sub>50</sub>	602	258	345	345
8	RC30G60	602	258	276	414
9	RC <sub>30</sub> G <sub>70</sub>	602	258	207	483
10	RC30G80	602	258	138	552
11	RC30G90	602	258	69	621
12	RC30G100	602	258	0	690

# Table 3-5: Test matrix

Specimen	Size (mm)	No.of	For	For	For	For	Total
		batch	Com.	tensile	Flexure	Flexure	no. of
			Strength	and	(28,56	28 days	sample
			(7,28,56	UPV	days)		
			days)	(28	3 point	4 point	
				days)			
Cylinder	100×200	15	135	45	-	-	180
Prism	100×100×	12	-	-	48	-	48
beam	500						
Beam	100×150×	12	-	-	-	24	24
	1100						

# 3.4 Workability test of concrete

The slump test is used to understand the workability of fresh concrete. Generally, higher slump indicates higher workability of concrete. Test procedure followed according to ASTM C143 standard. It is necessary to know the behavior of RBA based glass concrete in terms of workability. Besides, air content test indicates the amount of air in the concrete mix at the time of casting. Amount of air content which indicates the amount of void due to air. ASTM C231 is followed for this test. Fig. 3.3 shows the experimental test of the slump test of concrete mix and air content test.



Fig.3.3 (a) Slump test of Concrete mix (b) Air Content test of Concrete mix

# **3.5 Casting and Notation**

Casting is an important step of experimental research. Without proper casting and curing result of the research will not be satisfactory. So, in this research, every step of casting, notation, and curing has been done sincerely and by proper standards. Fig.3.4 shows the aggregates before mixing, doing proper vibration for reducing void, and finally doing casting and giving name of every specimen with date. According to this research plan, 180 cylinders and 48 prism beam has been cast.



Fig.3.4 (a-d) Mixing, Hammering, Casting and Notation



Fig.3.5 Strain gauge installation process of reinforced concrete (RC) beam



Fig.3.6 Notation of beams and cylinder

Another part of this thesis is to determine the flexural strength of the reinforced beam. As per plan 24 beam has been cast according to proper design. Fig.3.5 shows the installation process

of strain gauge in steel. After installation, it must be bound with waterproof tape. The outside portion of the wire should be in polythene to protect it from curing water and air. Fig.3.7 shows the curing process of the specimen. For cylindrical and prism beam specimens ponding curing has been chosen and for beam specimens jute shading is preferred.



Fig.3.7 Curing process of cylindrical, prism beam and reinforced beam specimen

# 3.6 Mechanical properties test of RCG

Generally, concrete is strong in compression and weak in tension. Compressive strength, Tensile strength, and hardened density are the major components of mechanical properties of concrete. The compressive strength test has followed the ASTM C39 standard. Besides splitting tensile strength has been done according to ASTM C496. Compression test of cylindrical specimen has been done for 7,28 and 56 days. The tension test of the cylindrical specimen has been done only for 28 days. Minimum three sample cast for each batch. In total 180 cylindrical has been prepared for testing. To know the behavior of recycled glass concrete it is necessary to do the proper test of mechanical properties.



Fig. 3.8: Compression test setup of Cylindrical specimen

## 3.7 Ultrasonic Pulse Velocity Test

The ultrasonic pulse velocity (UPV) method, as described in ASTM C597, is a nondestructive method used to determine the quality and integrity of concrete structures. It is performed by monitoring the transmission of elastic waves through concrete. An emitter transducer generates elastic waves in the concrete, which vibrate at their resonant frequency when short pulses of high-voltage electricity are sent. A receiver transducer detects these pulses placed nearby, and the time of transmission is determined by a device containing a timer connected to both transducers. From this, the pulse velocity, C, can be determined by the equation C = L/t, where L is the distance travelled and t is time. Three different configurations of transducers can be used to perform a UPV test. This includes direct, semi direct, and indirect (surface) transmission. Fig.3.9 shows the UPV test of the cylindrical specimen.



Fig 3.9: UPV test of cylindrical specimen

## 3.8 SEM test

The SEM (Scanning Electron Microscopy) images obtained using BSE imaging are used to quantify the microstructure features of the concrete. The microstructure properties will help determine the physical properties such as compressive, tensile, and flexural strength of the concrete. Generally concrete needs a coating to activate zero conductivity. Gold is perhaps the most widely used coating material for non-conductive SEM samples, but it is not recommended as a sputter coating for research purposes where high-magnification images are required. Gold has a high secondary electron yield and sputters relatively rapidly, but the coating structure is composed of large islands (grains) that can be observed at high magnification in most modern research levels. Several collected sample from the concrete cylinder has been set in the coating machine which is shown in Fig.3.10. After proper coating samples are set to SEM machine to get different scale and magnifications images to ensure the microstructure of Recycled Concrete containing Glass (RCG).



Fig.3.10 Coating on sample

# 3.9 Flexural test setup for prism beams

For flexural test huge number of prism beam has been ted. ASTM C78 standard has been followed which actually procedure of the third point is loading. Here total length of the beam reinforcement is not provided. The loading rate is 0.5mm per minute according to standard test protocol. The load has been inputted by the Universal Testing Machine (UTM) which loading capacity is 1000KN. Total 12 no. of batches has been cast. Test has been planned to do 28 days and 58 days. In total 48 prism beam has been casted for view flexural response of prism beam without reinforcement. Fig.3.11 shows the three-point bending test setup of prism beam.

Another part of flexural strength test was four-point loading test for reinforced beam (RC). In this part ASTM standard has been followed. Total length of the beam is 1100 mm where one end to support length is 100mm and span length is 900mm. Distance between two point loads is 200mm.A strain gauge is set in steel which is described before in casting part. Weld able strain gauges are installed on steel. It monitors the changes in load on structural elements during and after construction. Loading rate is about 1mm per minute according to standard

protocol. Load has been inputted by the Universal Testing Machine (UTM) which loading capacity is 1000 KN. A LVDT has been set at the mid-point of the beam. LVDT is an acronym for Linear Variable Differential Transformer. It is a common type of electromechanical transducer that can convert the rectilinear motion of an object to which it is coupled mechanically into a corresponding electrical signal. Fig.3.12 shows the loading setup of four-point bending test.



Fig.3.11 Three-point bending test setup of prism beam





Fig.3.12 (a,b) Four-point bending test setup of reinforced concrete (RC) beam

## 3.10 Summary

In this chapter, procedure of materials collection , casting and test methodos are described extensively. To achive the objectives of this research all test procedure has been occurred according to ASTM stanards. The key points of this chapter can be summarized as follows

- a) Recycled Brick (RB) is collected from a 30 years old demolished building and then proceed a standard form to use as coarse aggregate in concrete. Another core materials of this research is Waste glass (WG) which is collected from Puran Dhaka waste tube light processing factory.
- b) As Waste Glass (WG) is used as the replacement of fine aggregate. So, the particle distribution curve is within the limit of ASTM C33 limits after several sieving. Besides a number of basic test is occurred for deterimining the properties of aggregate.
- c) A proper mix design is prepared before casting procedure. For cyclindrical specimen there are fifteen batches and for prism beam and reinforced concrete (RC) beam contains twelve types of specimen .

- d) For determining workaibility of concrete slump and air content test is occurred according to ASTM C143 and ASTM C231 respectively. Besides compressive strength,splitting tensile strength and flexural test through center point loading is occurred according to ASTM C39,ASTMC496 and ASTM C293.
- e) Ultrasonic Pulse Velocity (UPV) of cylindrical specimen is occurred according ASTM C597. In addition Scanning Electron Microscopy (SEM) is occurred to get microstructure images of concrete.
- f) ASTM C78 standard is followed for four point flexural test of Reinforced Concrete (RC) beam. Linear Variable Differential Transducer (LVDT) is installed in the mid part and left mid part of the beam. Besides strain gauge is installed in steel before casting to get proper result.

#### **CHAPTER 4**

### FRESH AND HARDENED PROPERTIES OF RCG

#### **4.1 Introduction**

The fresh and hardened properties of RCG, including the air content test, slump test, compression, tensile, density, and flexural strengths of the prism beam, are discussed in this chapter. The findings of an ultrasonic pulse velocity (UPV) test and a scan from a scanning electron microscope (SEM) will also be briefly explained. Understanding the behavior of a new material type, such as waste tube glass, requires a comparison of this result to the normal specimen. There are other comparisons between compressive strength and density, UPV and compressive strength, compressive strength and tensile strength, and the fluctuation in compressive strength from day to day, which is between 7, 28, and 56 days. A variety of code comparisons, including those for CSA A23.3-14, ACI-318-14, and ACI-209-82, were also provided. A number of code comparisons like ACI-318-14, ACI-209-82, fib-2010, and CSA A23.3-14 were also included for splitting tensile strength.

### 4.2 Workability of RCG

Concrete workability describes how quickly and uniformly freshly mixed concrete can be laid, cemented, and completed. In general, how fluid the mix is affects how workable concrete is. A slump is a popular name for this. Basically, the more fluid the concrete, the higher the slump, and while the slump is thought of as a measure of water content, it is frequently also used as a measure of concrete consistency. Materials like water content, cement concrete, sand, and aggregate qualities like size, shape, grading, mix design ratio, and usage of admixtures are all factors that determine how workable concrete is. According to a series of experimental findings, replacing fine aggregate with glass improves workability compared to using regular aggregate [13, 14, 16-18, 51]. On the other hand, several researchers asserted that the slump value was considerably lower than what was obtained using sand as the control condition [14, 19-21, 51]. The difference in workability between the two categories of instances, nevertheless, is minimal.

The fifteen Recycled Concrete containing Glass (RCG) specimen slumps are depicted in Fig.4.1. Batch numbers 1 through 4 ( $RC_0G_0$  to  $RC_{30}G_0$ ) are only for samples of recycled concrete (RC) that include no waste glass at all. The slump has grown from batch no.2 to batch no.4 by around 2.82%, 4.2%, and 6.8% when compared to batch no.1. That indicates that using recycled brick material in concrete helps to increase its slump. With more recycled brick used

as coarse aggregate, the material is more workable overall. However, the slump has started to lessen for the addition of glass following batch no. 4 ( $RC_{30}G_0$ ). Slump decreased by roughly 6.8% and 11% for batches no. 5 and 6, respectively, in comparison to batch no. 4 ( $RC_{30}G_0$ ). Beginning with batch number seven ( $RC_{30}G_{25}$ ), the slump began to rise by around 3% in comparison to batch number four ( $RC_{30}G_0$ ), and the rise persisted through the last batch ( $RC_{30}G_{100}$ ). Compared to batch no. 6 ( $RC_{30}G_{20}$ ), the slump for ( $RC_{30}G_{50}$ ) has grown by almost 12%. Compared to batch no. 6 ( $RC_{30}G_{20}$ ), the slump for batch no. 15 ( $RC_{30}G_{100}$ ) has increased by roughly 21%. Up to 20% of glass replacement resulted in a decrease in fine aggregate workability, and from 20% to 100% of glass content, workability increased.



Fig. 4.1 Slump of RCG

The amount of air in concrete is also influenced by the glass sand content. Previous studies have shown that different concrete grades differ somewhat from one another. The air content is seen to decrease at a glass percentage of 25%, while the value increases with higher glass levels (Taha and Nounu, 2009). The air content of concrete with glass sand concentrations at 30%, 50%, and 70%, however, consistently increased by 12-41%, according to Park et al. (Park et al., 2004). The variation in air content is depicted in Fig. 4.2 as a percentage. The presence of air indicates emptiness and promotes improved workability. In comparison to batch no.1 (RC<sub>0</sub>G<sub>0</sub>), the air content has risen by roughly 0.31, 1.20, and 1.5 percent for batches no.2 (RC<sub>10</sub>G<sub>0</sub>), no.3, and no.4, respectively. However, the air content has started to diminish with the addition of glass as fine aggregate. Air content has been reduced by roughly 4.6% and 5.50% for the replacement of glass contents of 10% and 20%, respectively, when compared to

batch no. 4 ( $RC_{30}G_0$ ). In contrast to batch no. 6 ( $RC_{30}G_{20}$ ), batch no. 7 ( $RC_{30}G_{25}$ ) had a steady air content, while batch no. 8 ( $RC_{30}G_{30}$ ) saw a start to an increase in air content. Air content has been increased by 3.74% and 10% for batch no.10 ( $RC_{30}G_{50}$ ) and 15 ( $RC_{30}G_{100}$ ) respectively.



Fig. 4.2 Air Content of RCG

Table 4-1: Air content and Slump test result of RCG

Batch no.	Notation	Air Content	Slump (mm)
		(%)	
1.	$RC_0G_0$	3.22	69
2.	$RC_{10}G_0$	3.23	71
3.	RC20G0	3.26	72
4.	$RC_{30}G_{0}$	3.27	74
5.	RC30G10	3.12	69
6.	RC <sub>30</sub> G <sub>20</sub>	3.09	66
7.	RC <sub>30</sub> G <sub>25</sub>	3.09	68
8.	RC30G30	3.11	69
9.	RC <sub>30</sub> G <sub>40</sub>	3.15	72
10.	RC30G50	3.21	75
11.	RC30G60	3.27	76

12.	RC <sub>30</sub> G <sub>70</sub>	3.28	79
13.	$RC_{30}G_{80}$	3.32	80
14.	RC30G90	3.4	83
15.	RC <sub>30</sub> G <sub>100</sub>	3.43	84

#### **4.3 Compressive Strength**

One of the crucial mechanical characteristics to consider while evaluating concrete is its compressive strength. Results from tests on compressive strength are generally used to determine whether the concrete mixture delivered to the construction site satisfies the specifications for the specified strength, fc', in the job specification. The fabrication and curing of cylinders subjected to acceptance and quality control testing follow the same processes outlined in ASTM C-31 for standard-cured specimens. For field-cured specimens, ASTM C-31 offers methodologies for calculating the in-place concrete strength. As per ASTM C-39, cylindrical specimens are tested. In order to understand the real behavior of RCG, fifteen batches of recycled concrete glass (RCG) were examined for 7, 28, and 56 days.

The use of RG as a concrete replacement for the fine aggregate (RGFA) piqued the interest of numerous researchers. The majority of them demonstrated that the compressive strength of concrete can be significantly increased by substituting 20–30% RG for fine aggregate (mostly sand) (Jeba Samuel, 2018, Arivalagan and Sethuraman, 2021, Elavarasan, 2016, Pratheba.S, 2018, Rjoub and Tamimi, 2019, Alducin-Ochoa et al., 2021, J.Premalatha 2019, Ibrahim, 2020, Gautam et al., 2012). Furthermore, they asserted that the presence of RGA as a fine component of concrete does not significantly reduce compressive strength. However, as seen in numerous research works, the strength qualities may degrade if the replacement amount exceeds 30%. Additionally, compressive strength started to decrease after employing 30% of RGFA, which is also seen in the references (Ling et al., 2011, Herki et al., 2017, Ling and Poon, 2011, J.Premalatha 2019, Gerges et al., 2018, Corinaldesi et al., 2016, Tan and Du, 2013, Rahim et al., 2014). This phenomenon can be explained by the fact that fine glass can fill in gaps in the aggregate at a relatively lower percentage, increasing the overall compressive strength. On the other hand, due of the fine glass' smoother surface, more glass content in concrete results in smaller bonding between the fine content, which results in lower compressive strength.

The compressive strength of RCG samples varied throughout 7 days, as seen in Fig.4.3. As usual, the strength of the aggregate decreased as the percentage of recycled brick aggregate

increased. In comparison to batch 1 ( $RC_0G_0$ ), the strength of batch 2 ( $RC_{10}G_0$ ) through 4 ( $RC_{30}G_0$ ) fell by roughly 0.94 percent, 2.3%, and 7.5%, respectively. The strength trend began to shift after using glass content rather than sand. In comparison to batch no. 4 ( $RC_{30}G_0$ ), the strength of batch no. 5 ( $RC_{30}G_{10}$ ) and 6 ( $RC_{30}G_{20}$ ) has grown by roughly 7.4% and 12.5%, respectively. But strength started to decline once more starting with batch number 7 ( $RC_{30}G_{25}$ ). This indicates that the strength of the replaced glass has grown by up to 20% before beginning to decline. Strength has decreased by between 27.4% and 72.4% for sand replacement rates of 50% ( $RC_{30}G_{50}$ ) and 100% ( $RC_{30}G_{100}$ ) compared to batch no. 6 ( $RC_{30}G_{20}$ ).



Fig. 4.3 Compressive Strength of RCG (7 days)

The compressive strength changes over fifteen batches throughout 28 days is shown in Fig. 4.4. As usual, every batch's concept strength improved in comparison to the 7-day strength result. Even the pattern of shifting strength is very comparable to that of seven days' strength. Strength has decreased by about 12% when recycled brick aggregate ( $RC_{30}G_0$ ) is added compared to unrecycled brick aggregate ( $RC_0G_0$ ). The strength of batch no. 4 ( $RC_{30}G_0$ ) has been enhanced by roughly 15.5% to add glass content of about 20% ( $RC_{30}G_{20}$ ). Strength of the substituted glass began to decline at 25% ( $RC_{30}G_{25}$ ), which is approximately 1.2% less than batch no. 6 ( $RC_{30}G_{20}$ ). About 25 to 30 MPa was the design strength we aimed for after 28 days. Seven batches in total have satisfied this design requirement, two of which comprise just recycled aggregate with replacement percentages of around 10% and 20%, and the other four

of which contain recycled aggregate with a replacement percentage of about 30% and glass content ranging from 10% to 30%.

The exact value of compressive strength for fifteen batches over 7, 28, and 56 days is displayed in Table 4-2. Strength has increased as usual with the lengthening days. Nearly identical data are presented graphically in Fig.4.5. Compared to the strength result after 7 days, strength improved by 20% to 30% in 28 days. After that, strength grew by roughly 10% in 56 days compared to the test result after 28 days. Another two batches are included in accordance with the targeted strength we have set here. That indicates that out of 15 batches, a total of 9 batches have passed the 56-day strength test. Batch numbers 4 (RC<sub>30</sub>G<sub>0</sub>) and 9 (RC<sub>30</sub>G<sub>40</sub>) have been included as new batches within the scope of this research.



Fig. 4.4 Compressive Strength of RCG (28 days)

Table 4-2: Compressive Strength of RCG (7,28,56 days)

Batch no.	Notation	7 days	28 days	56 days
		strength	strength	strength
		(MPa)	(MPa)	(MPa)
1.	$RC_0G_0$	20.293	27.801	30.858
2.	$RC_{10}G_0$	20.103	26.743	29.682
3.	$RC_{20}G_0$	19.821	26.222	29.211

4.	$RC_{30}G_0$	18.782	24.412	27.045
5.	RC <sub>30</sub> G <sub>10</sub>	20.274	27.332	30.327
6.	RC30G20	21.453	28.881	31.769
7.	RC <sub>30</sub> G <sub>25</sub>	21.366	28.543	31.683
8.	RC30G30	20.678	27.433	30.379
9.	RC <sub>30</sub> G <sub>40</sub>	17.456	23.113	25.586
10.	RC30G50	15.592	20.732	22.834
11.	RC30G60	15.237	19.431	21.376
12.	RC30G70	13.252	16.895	18.586
13.	RC30G80	9.883	11.526	12.677
14.	RC <sub>30</sub> G <sub>90</sub>	7.868	8.883	9.772
15.	RC30G100	5.924	7.912	8.702



Fig. 4.5 Compressive Strength of RCG (7,28,56 days)

## 4.4 Splitting Tensile Strength

Numerous research has been done to ascertain the tensile characteristics of concrete formed from recycled glass concrete (RGC) using split tensile tests. Tensile strength of concrete made using RG as fine aggregate (RGFA) is less than that of typical concrete, according to several studies one (Arivalagan and Sethuraman, 2021, Rjoub and Tamimi, 2019, Ammash et al.,

2009).. On the other hand, certain studies revealed that adding RGFA increased the value of tensile strength (Elavarasan, 2016, Pratheba.S, 2018, Lokesh et al., 2017).

The tensile strength of recycled concrete glass (RCG) after 28 days is shown in Fig. 4.6. The aggregate strength of brick has decreased as the percentage of recycled brick has increased. Strength fell by almost 1.3%, 2.9%, and 5.26%, respectively, from batch no. 2 (RC<sub>10</sub>G<sub>0</sub>) to batch no. 4 (RC<sub>30</sub>G<sub>0</sub>), in comparison to batch no. 1 (RC<sub>0</sub>G<sub>0</sub>). However, as glass content was added, the strength of the fine aggregate began to rise. Glass additions of 10% and 20% each enhanced strength by around 0.5% and 1.5%, respectively. Strength rose even for batch no. 7 (RC<sub>30</sub>G<sub>25</sub>) compared to batch no. 4 (RC<sub>30</sub>G<sub>0</sub>) by almost 1.8%. Starting with batch no. 8 (RC<sub>30</sub>G<sub>30</sub>), strength began to decline. In comparison to batch no. 7 (RC<sub>30</sub>G<sub>25</sub>), the strength of fine aggregates with 50% and 100% glass content was decreased by nearly 14% and 68%, respectively.



Fig. 4.6. Splitting Tensile Strength of RCG

The values for compressive strength and tensile strength after 28 days are displayed in Table 4-3. The graphic representation of the 28-day strength is shown in Fig. 4.6. Tensile strength is roughly 11%, 10.8%, 10.7%, and 10.2% of the batch's compressive strength from batch no. 1 ( $RC_0G_0$ ) to batch no. 4 ( $RC_{30}G_0$ ). Tensile strength for batches no. 5 ( $RC_{30}G_{10}$ ) to no. 7 ( $RC_{30}G_{25}$ ) is approximately 11.4%, 12%, and 11.7% of compressive strength for those batches. Tensile strength for batches 10 and 15 was found to be 10% and 10.2%, respectively, of compressive strength after 28 days.

		28 days Compressive	28 days Tensile
Batch no.	Specimen	Strength (MPa)	Strength
			(MPa)
1.	$RC_0G_0$	27.801	2.524
2.	$RC_{10}G_0$	26.743	2.493
3.	RC <sub>20</sub> G <sub>0</sub>	26.222	2.452
4.	RC30G0	24.412	2.391
5.	$RC_{30}G_{10}$	27.332	2.4021
6.	RC30G20	28.881	2.426
7.	RC30G25	28.543	2.434
8.	$RC_{30}G_{30}$	27.433	2.3844
9.	RC30G40	23.113	2.125
10.	$RC_{30}G_{50}$	20.732	2.081
11.	RC30G60	19.431	1.882
12.	RC30G70	16.895	1.678
13.	RC30G80	11.526	1.221
14.	RC <sub>30</sub> G <sub>90</sub>	8.883	0.885
15.	$RC_{30}G_{100}$	7.912	0.778

# Table 4-3: Compressive and Tensile strength of RCG



Fig. 4.7. Compressive and Tensile strength (28 days)

#### **4.5** Flexural strength of Prism beam (without reinforcement)

To comprehend the flexural characteristics of concrete, bending strength tests must be conducted. The flexural strength of concrete made from recycled glass has been the subject of some prior research. Flexural strength is increased by up to 20% RGFA addition (Arivalagan and Sethuraman, 2021, Sunarsih et al., 2021). However, flexural strength only increased by 5% RGFA, according to another study (Lokesh et al., 2017). Some researchers shown that replacing up to 10% of the cement with recycled glass particles improves strength compared to using regular cement (Ubeid et al., 2020, Abbas et al., 2021).

Twelve batches in all were tested for flexural strength throughout 28 and 56 days. The outcome of the prism beam's flexural strength after 28 and 56 days is shown in Fig. 4.8. Here, the recycled brick proportion was changed from 0 to 100% while remaining constant at 30% from the previous example. For the replacement of glass up to 20% ( $RC_{30}G_{20}$ ), strength has risen by over 15% as compared to the standard one ( $RC_{30}G_{0}$ ). Strength began to decline from RC30G25, and by the time it reached 100% glass ( $RC_{30}G_{100}$ ), it had decreased by around 74%. Flexural strength results after 56 days also follow the same pattern as those after 28 days. As per typical, all batches have stronger results than the 28-day results. level has increased from 2% to 4% of the previous 28-day level.



Fig.4.8. Flexural Strength of RCG (28 and 56 days)

#### 4.6 Hardened Concrete Density

The density of hardened concrete is another factor to assess strength. A curved line in Fig.4.9 depicts the hardened density of fifteen batches. Here, density falls off as recycled brick aggregate content rises. By using 30% ( $RC_{30}G_0$ ) recycled brick instead of the original coarse aggregate, the strength has fallen by about 8%. However, increasing the amount of recycled glass to 20% ( $RC_{30}G_{20}$ ) revealed a value gain of roughly 10%. However, strength started to decline starting with  $RC_{30}G_{25}$  and reduced by over 41% for batch no. 15 ( $RC_{30}G_{100}$ ). The relationship between hardened density and compressive strength is depicted in Fig.4.10. Here, a nearly linear link has been established.



Fig. 4.9. Dry density of RCG



Fig. 4.10. Dry density Vs Compressive strength of RCG

#### 4.7 Ultrasonic Pulse Velocity

In structural engineering, ultrasonic measurements are used to analyze deterioration, find flaws, and determine material qualities. Velocity is a feature of ultrasonic wave propagation that can be applied in this situation. In this instance, cylindrical samples from fifteen different batches were evaluated 28 days following casting. After 28 days of casting, fifteen batches' UPV data are shown in Fig. 4.11. As the percentage of recycled brick aggregate rises, velocity falls. Strength falls for replacement recycled brick by 10%, 20%, and 30%, respectively, as compared to the fresh one. However, percentage velocity increased once glass was added. When 20% of the original glass is replaced ( $RC_{30}G_{20}$ ), as opposed to having no glass at all ( $RC_{30}G_{0}$ ), the strength increases by around 8.4%. In comparison to batch no. 6 ( $RC_{30}G_{20}$ ), velocity began to decline starting with batch no. 7 ( $RC_{30}G_{25}$ ). Strength dropped by 15% and 41%, respectively, for batches 10 ( $RC_{30}G_{50}$ ) and 15 ( $RC_{30}G_{100}$ ), compared to batch no. 6 ( $RC_{30}G_{20}$ ). A linear relationship between the results of the compressive strength test and UPV is shown in Fig. 4.12.



Fig. 4.11. Ultrasonic Pulse Velocity of RCG



Fig. 4.12. Relation between compressive strength and UPV

### 4.8 Comparison with the Standard Code of Practices

Foreseeing the mechanical response of concrete using its compressive strength is possible using a variety of design criteria. The different reaction parameters of the concrete specimens are predicted in the current study using design recommendations from the ACI, fib, and Euro-code (EC2), CSA. The formulae used to calculate the strength values are displayed in Table 4-4 and come from several codes. Tensile strength and flexural strength values derived from various codes are shown in Figs. 4.13 and 4.14, respectively. As shown in Figs. 4.15 and 4.16, the ratio (n) between experimental and predicted values is calculated for comparative purposes; a value of n larger than 1 denotes an over-prediction, whilst a value of n less than 1 denotes an underprediction of the outcome.


Fig. 4.13. Splitting Tensile strength values according to codes



Fig. 4.14. Flexural strength values according to codes



Fig. 4.15. Ratio between experimental and calculated values of Splitting Tensile strength



Fig. 4.16. Ratio between experimental and calculated values of Splitting Tensile strength

## 4.9 Failure Pattern of Concrete

The RCG concrete cylinder's failure pattern during the compressive strength test is depicted in Fig. 4.17. Here we have fifteen batches displayed. Cone and share failures are among the most typical failures. It has been found that increasing the glass content of concrete by up to 20%

improves bonding. It might occur as a result of the pozzolonic activity of glass. The failure pattern of (from left)  $RC_{30}G_{100}$ ,  $RC_{30}G_{30}$ , and  $RC_0G_0$  is depicted in Fig.4.18 (a). It has been noted that  $RC_{30}G_{100}$  exhibits one of the weakest bonds, while  $RC_{30}G_{30}$  exhibits a much stronger bond. The flexure failure of the  $RC_{30}G_{40}$  specimen, which is close to the middle of the beam, is shown in Fig.4.19. Due to flexure, a similar crack pattern is shown in Fig. 4.20. Twelve specimens fail not far from the beam's centre.



Fig. 4.17. Failure pattern of RCG concrete cylinder in compression



Fig. 4.18. (a) Failure pattern of RCG concrete cylinder in tension (b) cross section after failure



Fig. 4.19. Failure pattern of RC30G40 prism beam in flexure

RC 30 Gr up c/11/22	RC36 G750 6/ 11/22
RC30 G730 10 6/ 11/22	RC30 G1 60 0 6/11/22
RC30 64/21/22	RC 30 G170 6/11/22
RC30 G125 10 G14/22	RC 30 G180 6/11/22
RC30 6130 (4/11/22	RC30 6190 900 6111/2-
RC30 GP40- 30 9/40/22	RC 30 Gr 100 6/ 11/ 22

Fig. 4.20. Flexural failure pattern of RCG concrete

### 4.10 SEM images

Scanning Electron Microscope (SEM) examination enables us to confirm how alterations are created within the sense of the material and the interfaces added to the paste, sort of a severe presence of micro-cracks within the terrain of the grains.Fig.4.21. (a-b) shows the micrographic images of large scale after 56 days of casting for the batch no.15 ( $RC_{30}G_{100}$ ). Magnification was 659 and 2.3k and scale was 30µm and 10µm respectively. From strength result discussion it is clear that 100% replacement of sand by glass showed the lowest strength. Here lots of voids and crack and fissures are created. This may be happening lack of proper bonding with aggregates and cement and ASR reaction.Fig.4.22 shows the SEM images of  $RC_{30}G_{20}$  batch which has twenty percentage glass. It has some micro cracks but much better than  $RC_{30}G_{100}$  sample. Images has been taken 126 and 4k magnification and 200µm and 10µm scale respectively. As said according to strength result it gained highest strength among RCG batches. Micrographic images also indicate the lesser voids and cracks which indicates the higher strength. Due to pozzolonic action of glass it maybe happened.



Fig. 4.21. SEM images of  $RC_{30}G_{100}$  sample (a)  $30\mu m$  (b)  $10\mu m$ 





Fig. 4.22. SEM images of  $RC_{30}G_{20}$  (a) 200 $\mu$ m (b) 10 $\mu$ m

## 4.11 Summary

In this chapter lab result of RCG especially fresh and hardened properties has been showed elaborately. Several comparisons also showed to understand the actual behavior of RCG. The test results of this chapter can be summarized below

- a) Workability reduces for waste glass content replacement up to 20% (RC<sub>30</sub>G<sub>20</sub>). It then started to grow after that. The amount, kind, and shape of glass particles have an impact on how easily RCG concrete can be worked.
- b) In terms of compressive strength, as usual strength decreases with the increasing percentage of recycled brick aggregate. Adding glass content up to 20% gives 12.5% more strength. But after that strength decreases with the increasing amount of glass content. Maybe for recycled tube glass, 20% is the optimum level in concrete. Almost similar trend has been followed in terms of the tensile strength of RCG. But here glass content of up to 25% (RC<sub>30</sub>G<sub>25</sub>) showed an increasing trend of strength.
- c) Hardened density decreases with the increasing percentage of recycled brick aggregate. But for adding glass up to 20% density increases by almost 10% compared to the fresh one. Compressive strength maintains almost a linear relationship with hardened density.
- d) In terms of Ultrasonic pulse velocity, for adding glass content up to 20% showed an increasing trend. Almost 8.4% velocity increases due to adding glass content up to 20%. Besides SEM result shows the larger voids and fissures for adding more glass content. But for up to 20% glass content SEM showed denser structure, lesser voids, and fissures. Maybe for the Alkali-Silica Reaction it happened.
- e) In terms of the flexural strength of the prism beam, strength increases for adding glass content up to 20% and after that starts to decrease. Almost 15% strength has been increased by adding 20% glass content.
- f) Failure pattern of cylinder specimen has no huge difference for changing glass content. The most common cone and shear failure occurred. Besides crack pattern of prism beam is as usual flexural crack.

#### **CHAPTER 5**

### FLEXURAL RESPONSE OF REINFORCED CONCRETE BEAM

### **5.1 Introduction**

Till now, reinforced concrete structures have been the most popular choice of civil and structural engineers. They are by far considered the most successful construction materials because of their versatility, robustness, economic feasibility, and durability. This Chapter contains flexural response of reinforced concrete beam through load-deflection chracteristics, strain of steel, crack pattern of reinforced concrete beam and flexural strength of beam. Twelve varieties of reinforced concrete beam specimen is ready for the test. Load-Deflection data is recording through LVDT,UTM and also load-strain data is recording from starin gauge of steel. Besides no.of crack and pattern of crack on beam is noted in time of testing. Experimental findings have led to determinations of cracking, yielding, and ultimate moment capacity, which have been compared to the ACI-318-14 code.

### **5.2 Load-Deflection Characteristics**

Fig.5.1 shows the load versus deflection responses at the midpoint of the tested RCG beams. Experimental results show that the load-deformation behavior of RCG beams is slightly different. For the  $RC_{30}G_0$  beam, the highest load has been taken which is 58 kN with a displacement near to 6mm. After that, due to adding glass content, the slope of the load-deflection curve starts to decrease gradually with an increasing number of cracks developed in the tested beam specimens. With further increase in the applied load, the longitudinal reinforcements start to yield and then the slope of the curves significantly declines beyond the yielding point. In this stage, with a small increment in the applied load, the deflection suddenly increases and reaches the maximum load-carrying capacity when the concrete crushing takes place with an abrupt drop in the load level. On the other hand, from the load-deflection response of  $RC_{30}G_{10}$ ,  $RC_{30}G_{20}$ , and  $RC_{30}G_{25}$  beams, it is observed that the ability to take load decreases with the increasing amount of glass which is 7.7%%,7.8%, and 8.7% respectively. After reaching the ultimate capacity the load drops abruptly. For  $RC_{30}G_{50}$  and  $RC_{30}G_{100}$  load carrying capacity decreases by almost 15% and 32% respectively compared to the fresh one ( $RC_{30}G_{0}$ ). In terms of displacement adding glass content doesn't create a significant effect on concrete.





Fig.5.1 Load - Deflection behavior of RC beam

Fig.5.2 shows load and deflection data from LVDT for mid and left portions. As discussed in the methodology chapter test setup part two LVDTs have been set in times of testing. One is under the midpoint of the beam span and another one is left mid part of the beam. Almost all specimen indicates that the major failure was done by flexure and then shear failure also occurred sometime. Ultimate load carrying capacity has been decreased with the increasing percentage of glass content. Maximum deflection occurs at the mid-point or near to mid-point. Experimental results show that the load-deformation behavior of RCG beams is slightly different. For the RC<sub>30</sub>G<sub>0</sub> beam, the highest load has been taken which is 62.5 kN with a displacement near 6.79mm at the mid-point. After that, due to adding glass content, the loadcarrying capacity started to decrease gradually with an increasing number of cracks developed in the tested beam specimens. With further increase in the applied load, the longitudinal reinforcements start to yield and then the slope of the curves significantly declines beyond the yielding point. In this stage, with a small increment in the applied load, the deflection suddenly increases and reaches up to the maximum load-carrying capacity when the concrete crushing takes place with an abrupt drop in the load level. On the other hand, from the LVDT data RC30G10, RC30G20, and RC30G25 beams, it is observed that the ability to take load decreases with the increasing amount of glass which is 1.76%, 2.56%, and 10% respectively. For RC<sub>30</sub>G<sub>50</sub>



and  $RC_{30}G_{100}$  load carrying capacity decreases by almost 17% and 43% respectively compared to the fresh one ( $RC_{30}G_0$ ).



Fig. 5.2 Deflections from LVDT

Fig.5.3 shows load and steel strain data from the strain gauge of steel. As discussed in the methodology chapter test setup a strain gauge has been set in the mid-down portion of the

beam. Here Ultimate load carrying capacity has been decreased with the increasing percentage of glass content. Experimental results show that the load-strain behavior of RCG beams is slightly different. For the  $RC_{30}G_0$  beam, the highest load has been taken which is 62.57 kN with a strain near to 0.004. That means steel crossed its yield limit. After that, due to adding glass content, the load-carrying capacity decreased gradually with an increasing number of cracks developed in the tested beam specimens. With further increase in the applied load, the longitudinal reinforcements start to yield and then the slope of the curves significantly declines beyond the yielding point. On the other hand, from the Strain gauge data  $RC_{30}G_{10}$ ,  $RC_{30}G_{20}$ , and  $RC_{30}G_{25}$  beams, it is observed that the ability to take load decreases with the increasing amount of glass which is 2.81%,3.03%, and 11% respectively. For  $RC_{30}G_{50}$  and  $RC_{30}G_{100}$  load carrying capacity decreases by almost 15% and 38% respectively compared to the fresh one ( $RC_{30}G_{0}$ ).







Fig 5.3. Strain Gauge data (Load Vs Strain)

# 5.3 Crack Pattern of RC Beam

The structural design of this beam target was to ensure the crack flexure fast and after some time finally shear crack also formed. Fig.5.4 shows the crack formation of  $RC_{30}G_{80}$  specimen. Cracks no. are set serially with time in the time of formation. The first series of cracks has been marked as flexural cracks and at the last some shear cracks also formed. More or less crack pattern is almost similar for all the batches for the RCG specimen. There is no huge difference in cracks due to changing glass content. Almost similar cracks have occurred in all samples. Fig.5.5 shows the crack pattern of twelve batches.



Fig.5.4 Crack Pattern of RC30G80 beam



Fig.5.5 Crack of RCG beam Specimen

# 5.4 Flexural Strength of RCG beam

As per ASTM C 78 flexural strength of RCG beam has been done. From the experimental result of UTM and written data at the time of testing Cracking load, yielding load, and Ultimate load have been noted. From there moment is calculated. Table 5-1 shows the value of moment capacity in cracking, yielding, and ultimate according to their batch. Reinforcement is used as per the requirement of moment and shear. Here the variation of flexural capacity changing trend is quite different from the prism beam specimen. Flexural capacity decreased from the beginning of adding glass content. Ultimate capacity has decreased by 8.5%, 9%, and 10% for the replacement of glass content 10%, 20% and 25% respectively. For batch no.7 (RC<sub>30</sub>G<sub>50</sub>) strength has decreased 16% compared to the fresh one. Even for batch no.12 (RC<sub>30</sub>G<sub>100</sub>) strength has decreased by 33.92% compared to batch no.1. Fig.5.6 shows the trend of changing the ultimate flexural capacity of the RCG beam. Overall a decreasing trend has been shown for adding glass. Maybe the bonding between reinforcement and glass content has an effect on overall strength performance.

Table 5-2 shows the equation for calculating theoretical moment capacity according to ACI-318-14. Fig.5.7 reports the ratio of the ultimate experimental moment capacity to the ultimate moment capacity recommended by the ACI-318-14 code. The comparison reveals that the experimental ultimate moment capacity for RCG beams is near to the predicted moment capacity obtained from the ACI code. This difference may arise from the fluctuation of design concrete and steel strength from the actual strength.

Specimen		Force, (kN)		Moment Capacity, (kN-m)		
	Cracking, Pcr	Yielding, Pv	Ultimate, Pu	Cracking, Mcr	, Yielding, My	Ultimate, Mu
RC <sub>30</sub> G <sub>0</sub>	17.49	43.77	58.29	3.06	7.65	10.20
RC <sub>30</sub> G <sub>10</sub>	14.92	39.88	53.32	2.61	6.97	9.33
RC <sub>30</sub> G <sub>20</sub>	14.80	39.53	52.70	2.59	6.91	9.22
RC30G25	14.76	39.32	52.5	2.58	6.88	9.18
RC <sub>30</sub> G <sub>30</sub>	13.98	38.01	50.73	2.44	6.65	8.80
RC30G40	13.70	36.78	49.05	2.39	6.43	8.58
RC30G50	13.56	36.51	48.4	2.37	6.38	8.47
RC30G60	13.56	36.31	48.3	2.37	6.35	8.45
RC <sub>30</sub> G <sub>70</sub>	13.30	35.41	47.22	2.32	6.19	8.26
RC <sub>30</sub> G <sub>80</sub>	12.75	34.11	45.48	2.23	5.96	7.95
RC <sub>30</sub> G <sub>90</sub>	12.22	33.27	44.38	2.13	5.82	7.76
RC <sub>30</sub> G <sub>100</sub>	10.8	28.22	38.56	1.89	4.93	6.74

Table 5-1: Flexural capacity of RCG beam

Table 5-2: Ultimate moment capacity equation (ACI-318-14)

Code	Year	Equation for Mu	
		$\rho_s = \frac{A_s}{bd}, \alpha = 0.85$	
ACI-318	2014	$M_u = \rho_s f_s b d^2 (1 - \frac{\rho_s f_s}{2 \times \alpha_1 f_c})$	







Fig.5.7 Comparison of Ultimate moment capacity of RCG beam with ACI-318 (2014)

# 5.5 Summary

The flexural response of the RCG beam is the main topic of discussion in this chapter. That was the last objective of this study. This data was obtained using LVDT data that was set in the mid and left-mid of the beam. A number of data points from LVDTs have been given, including displacement with distance and load changes. Additionally, the UTM data has demonstrated the behavior of the beam's load deflection. In this chapter, steel strain gauge data

is also examined. From the commencement of the glass addition, strength began to decline in this section. This chapter's synopsis can be expressed as follows:

- a) For the  $RC_{30}G_0$  beam, the highest load has been taken which is 58 kN with a displacement near to 6mm based on UTM data. After that, due to adding glass content, the slope of the load-deflection curve starts to decrease gradually with an increasing number of cracks developed in the tested beam specimens. The bonding between the reinforcement and glass content may affect on strength.
- b) For the  $RC_{30}G_0$  beam, the highest load has been taken which is 62.5 kN with a displacement near 6.79mm at the mid-point based on LVDT data. After that, due to adding glass content, the load-carrying capacity started to decrease gradually with an increasing number of cracks developed in the tested beam specimens.
- c) Experimental results show that the load-strain behavior of RCG beams is slightly different. For the  $RC_{30}G_0$  beam, the highest load has been taken which is 62.57 kN with a strain near to  $4000 \times 10^{-6}$  based on strain gauge data. After that, due to adding glass content, the load-carrying capacity decreased gradually with an increasing number of cracks developed in the tested beam specimens. Most of the failure showed that steel reaches its yield point and then cracks the concrete.
- d) In terms of LVDT, UTM, and strain gauge data, the load carrying capacity of RCG is reduced by 1%-8% for adding glass content up to 20% replacement. The bonding between reinforcement and glass has an effect on flexural strength.
- e) In terms of the crack pattern most of the cracks are in the flexural zone and some cracks are the combination of flexural and shear crack.
- f) In terms of code comparison, the experimental ultimate moment capacity for RCG beams is near the predicted moment capacity obtained from the ACI-318-14 code. Some differences may arise from the fluctuation of design concrete and steel strength from the actual strength.

#### **CHAPTER 6**

### CONCLUSIONS AND RECOMMENDATIONS

### 6.1 General

A mix design of a total of fifteen different mixtures is planned with a comprehensive test matrix. To fulfill the first two objectives of this research a series of experimental investigations are conducted to determine the physical mechanical and strength properties of RBA-based waste glass concrete. The physical properties of concrete such as specific gravity, workability, fineness, flow, air entrainment, and permeability tests are conducted following respective ASTM guidelines. After assessing the basic physical properties, cylindrical specimens are prepared to conduct compression, split tensile, and flexural behaviour of RBC with glass following ASTM C39, ASTM C496, and ASTM C78 respectively. A batch of control specimens is prepared and tested to compare the properties of recycled batches. The strength tests are performed at 7 days, 28 days, and 56 days. In addition, UPV tests have been conducted on each sample of different batches. Microstructural properties of RBA based WG concrete are also examined. Finally, some reinforced beam specimens are prepared for further bending tests to understand the behaviour of reinforced concrete beams following ASTM C 293 ultimately demonstrating the potential structural application of concrete containing RBCA and WGP.

## **6.2** Conclusions

Based on the conducted experimental investigations, the key findings of this study are as follows

- a) Workability was seen to be reduced with the increased percentage of glass content (up to 20%) (RC<sub>30</sub>G<sub>20</sub>). It then started to grow after that. The amount, types, and shape of glass particles have impacts on the workability and flowability of RCG concrete.
- b) Adding glass content up to 20% provides 12.5% higher compressive strength than that of without glass. However, the strength decreases with the further addition of glass content in the mix. Similar trends have been followed for the split tensile strength of RCG.
- c) For the addition of glass up to 20%, an increment of density was observed as high as 10% compared to that of the control specimens.
- d) According to Scanning Electron Micrographs (SEM) larger voids and fissures are shown as glass content increases. However, SEM showed a denser structure, fewer

voids, and fissures up to 20% glass content. It is possible to propagate the Alkali-Silica Reaction in the hardened concrete.

e) For the  $RC_{30}G_0$  beam, the highest load has been taken which is 62.57 kN with a strain near to  $4000 \times 10^{-6}$  based on strain gauge data. After that, due to adding glass content, the load-carrying capacity decreased gradually with an increasing number of cracks developed in the tested beam specimens.

## **6.3 Recommendations for Future Research**

Based on the literature review and this research there is more scope of work on RCG concrete. Recommendations for future research can be:

- a) Based on size, shape, texture and type of glass strength can be different for RCG. Fresh and hardened properties of concrete may vary with those properties. Those effects can be scrutinized for a wide variety of applications of waste glasses.
- b) Due to some pozzolanic action of glass, it can be used as a replacement of cement up to a specific percentage. The production process of cement affects our environment and a replacement may contribute to the sustainability of the environment.
- c) The effect of superplasticizers and admixtures on RG concrete can be an interesting topic to investigate. The application of admixture may open the window to add a higher percentage of waste glass in concrete with desired mechanical properties.
- d) The application of RG concrete in concrete structures has not been investigated yet. The flexural and shear behavior of concrete beams or the axial response of concrete columns containing RG concrete has not been investigated comprehensively.
- e) Using the existing test data, artificial intelligence can be another option to figure out the optimum percentage of RG for the best mechanical properties. In addition, individual ingredient dependencies of the strength parameter through machine learning can be a good scope of research.
- f) The durability of RG concrete is a major concern as there is a major concern of alkalisilica reaction. Weather effects like salinity and snowfall on RG concrete can be a research opportunity.

### REFERENCES

- Abbas, Z., Mahadi, H. & Tayeh, B. (2021). Producing Sustainable Concrete using Nano Recycled Glass. *The Open Civil Engineering Journal*, 15, pp. 236-243.
- Agwa, I. S., Omar, O. M., Tayeh, B. A. & Abdel Salam, B. A. (2020). Effects of using rice straw and cotton stalk ashes on the properties of lightweight self-compacting concrete. *Construction and Building Materials*, 235, 117541.
- Al-Akhras, N. M. (2006). Durability of metakaolin concrete to sulfate attack. *Cement and Concrete Research*, 36, pp. 1727-1734.
- Al-Mutairi, N., Terro, M. & Al-Khaleffi, A.-L. (2004). Effect of recycling hospital ash on the compressive properties of concrete: statistical assessment and predicting model. *Building and Environment*, 39, pp. 557-566.
- Al-Zubaidi, A. B., Shabeeb, K. M. & Jan, A. I. A. (2017). A Study of the Effect of Recycled Mix Glass on the Mechanical Properties of Green Concrete. *Journal of University of Babylon for Engineering Sciences*, 26, pp. 265 - 272.
- Al Saffar, D. (2017). Experimental Investigation of Using Ultra-Fine Glass Powder in Concrete. 7, pp. 33-39.
- Alducin-Ochoa, J. M., Martin-Del-Rio, J. J., Torres-Gonzalez, M., Flores-Ales, V. & Hernandez-Cruz, D. (2021). Performance of mortars based on recycled glass as aggregate by accelerated decay tests (ADT). *Construction and Building Materials*, 300, 124057.
- Ammash, H., Muhammad, S. & Nahhab, A. (2009). Using of waste glass as fine aggregate in concrete. *Al-Qadisiya Journal For Engineering Sciences*, 2, pp. 206-214.
- Aper Zava, S. N. A., Paul Terkumbur Adeke (2020). The Optimum Amount of Waste Glass Aggregate that can Substitute Fine Aggregate in Concrete. *Aksaray J.Sci.Eng*, 4, pp. 159-171.
- Arivalagan, S. & Sethuraman, V. S. (2021). Experimental study on the mechanical properties of concrete by partial replacement of glass powder as fine aggregate: An environmental friendly approach. *Materials Today: Proceedings*, 45, pp. 6035-6041.
- Bazhuni, M. F., Kamali, M. & Ghahrimaninezad, A. (2019). An investigation into the properties of ternary and binary cement pastes containing glass powder. *Frontiers of Structural and Civil Engineering*, 13, pp. 741-750.
- Bentchikou, M., Bouriah, S., Hamdane, N., Debieb, F. & Boukendakdji, O. (2017). Comparative study of the properties of mortars with recycled glass aggregates incorporated by addition and substitution. *Energy Procedia*, 139, pp. 499-504.
- Bostanci, S. C. (2019). Engineering Properties and Sustainability Assessment of Recycled Glass Sand Concrete. *European Journal of Science and Technology*, 17, pp. 117-130.

- Corinaldesi, V., Nardinocchi, A. & Donnini, J. (2016). Reuse of recycled glass in mortar manufacturing. *European Journal of Environmental and Civil Engineering*, pp. 1-12.
- Czapik, P., Kuza, D. & Boron, M. (2021). Influence of the Waste Glass Uses on the Cement Mortar Properties. *Structure and Environment*, 13, pp. 45-51.
- Debska, B., Licholai, L. & Silva, G. J. B. (2020). Effects of waste glass as aggregate on the properties of resin composites. *Construction and Building Materials*, 258, 119632.
- Drzymala, T., Zegardlo, B. & Tofilo, P. (2020). Properties of Concrete Containing Recycled Glass Aggregates Produced of Exploded Lighting Materials. *Materials (Basel, Switzerland)*, 13.
- Du, H. & Tan, K. (2014). Concrete with Recycled Glass as Fine Aggregates. *Aci Materials Journal*, 111, p. 47-58.
- Du, H. & Tan, K. H. (2013). Use of waste glass as sand in mortar: Part II Alkali–silica reaction and mitigation methods. *Cement and Concrete Composites*, 35, p. 118-126.
- Duraman, S. B. & Li, Q. (2021). Recycled Glass Cullet as Fine Aggregate and Partial Cement Replacement in Concrete. *IOP Conference Series: Materials Science and Engineering*, 1144, 012010.
- Dvorak, K., Dolak, D. & Dobrovolny, P. (2017). The Improvement of the Pozzolanic Properties of Recycled Glass during the Production of Blended Portland Cements. *Procedia Engineering*, 180, p. 1229-1236.
- Elavarasan, D. (2016). Experimental Study on Waste Glass as a Partial Replacing Material in Concrete for Fine Aggregate. *International Journal of Advanced Research in Biology Engineering Science and Technology*, 2, p. 116-120.
- Fahad, B. M. & Ali, N. S. (2020). Replacement of Natural Gravel with Waste Glass in Concrete. *IOP Conference Series: Materials Science and Engineering*, 870, 012147.
- Gautam, S., Srivastava, V. & Agarwal, V. (2012). Use of glass wastes as fine aggregate in Concrete. *J Acad Indust Res*, 1.
- Gerges, N. N., Issa, C. A., Fawaz, S. A., Jabbour, J., Jriege, J. & Yacoub, A. (2018). Recycled Glass Concrete: Coarse and Fine Aggregates. *European Journal of Engineering and Technology Research*, 3, p. 1-9.
- Hamid, R., And M. A. Zubir (2014). The Flexural Properties of Reinforced Recycled Glass Concrete Beam. *Materials Science Forum*, 803, p. 325–329.
- Haramkar, P. N., S. M., R. M. K., R. R. S., J. R. M. & V. K. (2018). Partial Replacement Of Fine Aggregate By Using Glass Powder. *International Conference* on Emanations in Mordern Engineering Science & Management (ICEMESM-2018).
- Harrison, E., Berenjian, A. & Seifan, M. (2020). Recycling of waste glass as aggregate in cement-based materials. *Environmental Science and Ecotechnology*, 4, 100064.

- Hasani, H., , A. S., , H. R. N., Mojtaba Khosravi, Zadanbeh & , M. B. A. (2022). A Comprehensive Review of Utilizing Crushed Waste Glass as Green Construction Material. 2nd International Conference on Architecture, Civil Engineering, Urban Development, Environment and Horizons of Islamic Art
- Hendi, A., Mostofinejad, D., Sedhaghatdost, A., Zohrabi, M., Naeimi, N. & Tavakolinia, A. (2019). Mix design of the green self-consolidating concrete: Incorporating the waste glass powder. *Construction and Building Materials*, 199, p. 369-384.
- Hendi, A., Rahmani, H., Mostofinejad, D., Tavacolinia, A. & Khosravi, M. (2017). Simultaneous effects of microsilica and nanosilica on self-consolidating concrete in a sulfuric acid medium. *Construction and Building Materials*, 152, p. 192-205.
- Heriyantho, Pahlevani, F. & Sahajwala, V. (2018). From waste glass to building materials An innovative sustainable solution for waste glass. *Journal of Cleaner Production*, 191, p. 192-206.
- Herki, B., Sherwani, A. F., Safari, Z., Mohammad, S. & Jawhar, Y. (2017). Promoting the Use of Waste Glass Concrete in Developing Countries. *International Journal of Science Education*, 8, p. 121-125.
- Huseien, G. F., Hamzah, H. K., Mohd, A. R., Khalid, N. H. A., Shah, K. W., Deogrescu, D. P. & Mirza, J. (2020). Alkali-activated mortars blended with glass bottle waste nano powder: Environmental benefit and sustainability. *Journal of Cleaner Production*, 243, 118636.
- Ibrahim, K. (2020). Recycled Waste Glass [WG] in Concrete. *Global Journal of Engineering Sciences*, 6(1): 2020., p. 1-8.
- Idir, R., Cyr, M. & Pavoine, A. (2020). Investigations on the durability of alkali-activated recycled glass. *Construction and Building Materials*, 236, 117477.
- Islam, G. M. S., Rahman, M. H. & Kazi, N. (2017). Waste glass powder as partial replacement of cement for sustainable concrete practice. *International Journal of Sustainable Built Environment*, 6, p. 37-44.
- J. Rindl, D. C. (1998). Report of Dept. of Public Works. Madison, USA.
- J.Premeltha, R. S. (2019). Properties of Concrete with Waste Glass Powder (GP) as Fine Aggregate Replacement. *International Journal of Recent Technology and Engineering* 8.
- Jabal, Q. A., Al-Baghdadi, W. H. & Al-Hussaini, O. M. (2021). Mechanical properties and durability for concrete using waste heat resistance glass (Pyrex Type) as coarse aggregate. *Journal of Physics: Conference Series*, 1973, 012171.
- Jeba Samuel, K. L. R. (2018). Experimental Study on Partial Replacement of Fine Aggregate by Waste Glass Powder and CNC Lathe Waste. *IJSRD - International Journal for Scientific Research & Development*/, 6.

- Jia, S. L., Gong, H., Zhang, B. C., Zhang, C. F. & LI, Y. Y. (2015). Research on Waste Glass Concrete Elasticity Modulus. *Applied Mechanics and Materials*, 744-746, p. 1477-1480.
- Kalakada, Z. & Doh, J. H. (2020). Studies on Recycled Waste Glass Powder as Binder in Concrete.
- Kalifa, P., Menneteau, F.-D. & Quenard, D. (2000). Spalling and pore pressure in HPC at high temperatures. *Cement and Concrete Research*, 30, p. 1915-1927.
- Khalil, A. S., Mahmoud, M. A., Al-Hathal, A., Jawad, M. K. & Mozahim, B. M. (2018). The effect of replaced recycled glass on thermal conductivity and compression properties of cement. *Journal of Physics: Conference Series*, 1003, 012127.
- Kim, K.-H., Jeon, S.-E., Kim, J.-K. & Yang, S. (2003). An experimental study on thermal conductivity of concrete. *Cement and Concrete Research*, 33, p. 363-371.
- Kou, S. C. & Poon, C. S. (2009). Properties of self-compacting concrete prepared with recycled glass aggregate. *Cement and Concrete Composites*, 31, p. 107-113.
- Li, A., Qiao, H., Li, Q., Hakujwezu, T. & Chen, B. (2021). Study on the performance of pervious concrete mixed with waste glass powder. *Construction and Building Materials*, 300, 123997.
- Limbachiya, M. C. (2009). Bulk engineering and durability properties of washed glass sand concrete. *Construction and Building Materials*, 23, p. 1078-1083.
- Ling, T.-C. & Poon, C.-S. (2011). Properties of architectural mortar prepared with recycled glass with different particle sizes. *Materials & Design*, 32, p. 2675-2684.
- Ling, T.-C., Poon, C.-S. & Kou, S.-C. (2011). Feasibility of using recycled glass in architectural cement mortars. *Cement and Concrete Composites*, 33, p. 848-854.
- Liu, S., Wang, S., Tang, W., Hu, N. & Wei, J. (2015). Inhibitory Effect of Waste Glass Powder on ASR Expansion Induced by Waste Glass Aggregate. *Materials (Basel)*, 8, p. 6849-6862.
- Lokesh, S., Vaisya, P. & Chandrasekharan, P. (2017). Study on Mechanical Properties of Concrete by Partial Replacement of Crushed Glass as Fine Aggregate with M 20 Grade. <u>www.irjet.net</u> 4.
- Mallum, I., Mohd.Sam, A. R., Lim, N. H. A. S. & Omolayo, N. (2022). Sustainable Utilization of Waste Glass in Concrete: a Review. *Silicon*, 14, p. 3199-3214.
- Mishra, S., Thakur, K. & Gupta, V. (2020). Partial Replacement of Fine Aggregate by Glass Powder in Concrete. *International Journal of Engineering Research and*, V9.
- Nasr, M., Salih, S. & Hassan, M. (2016). Some Durability Characteristics of Micro Silica and Nano Silica Contained Concrete. *Journal of Babylon University*, 24, p. 980-990.

- Nwofor, T. & Ukpaka, C. (2017). Predictive Models for Compressive Strength of Concrete Produced with Waste Glass as Replacement for Aggregates. *International Journal of Civil Engineering*, 4, p. 31-34.
- Özkan, Ö. & Yuksel, İ. (2008). Studies on mortars containing waste bottle glass and industrial by-products. *Construction and Building Materials*, 22, p. 1288-1298.
- Park, S. B., Lee, B. C. & Kim, J. H. (2004). Studies on mechanical properties of concrete containing waste glass aggregate. *Cement and Concrete Research*, 34, p. 2181-2189.
- Rahim, N. L., Che Amat, R., Mohammad Ibrahim, N., Salehuddin, S., Mohammed, S. &
  Abdul Rahim, M. (2014). Utilization of Recycled Glass Waste as Partial Replacement of Fine Aggregate in Concrete Production. *Materials Science Forum*, 803, p. 16-20.
- Renga Rao Krisnamoorthy, J. A. Z. (2013). Thermal Conductivity and Microstructure of Concrete Using Recycle Glass as a Fine Aggregate Replacement. <u>www.ijetae.com</u> 3.
- Scharz, N., Cam, H. & Neithalath, N. (2008). Influence of a fine glass powder on the durability characteristics of concrete and its comparison to fly ash. *Cement and Concrete Composites*, 30, p. 486-496.
- Sharifi, Y., Afshoon, I. & Firozaiee, Z. (2015). Fresh Properties of Self-Compacting Concrete Containing Ground Waste Glass Microparticles as Cementing Material. *Journal of Advanced Concrete Technology*, 13, p. 50-66.
- Shin, A. H.-C. & Kodide, U. (2012). Thermal conductivity of ternary mixtures for concrete pavements. *Cement and Concrete Composites*, 34, p. 575-582.
- Sopov, V. P., Korkh, O. I. & Izbash, M. Y. (2020). A study of the alkali-silica reaction in recycled glass concrete. *IOP Conference Series: Materials Science and Engineering*, 907, 012062.
- Srivastava, V., Gautam, S., Agarwal, V. & Mehta, P. (2014). Glass Wastes as Coarse Aggregate in Concrete. *Journal of Environmental Nanotechnology*, 3, p. 67-71.
- Sunarsih, E. S., Patani, G. W., Agustin, R. S. & Rahmawati, K. (2021). Utilization of Waste Glass and Fly Ash as a Replacement of Material Concrete. *Journal of Physics: Conference Series*, 1808, 012004.
- Taha, B. & Nounu, G. (2009). Utilizing Waste Recycled Glass as Sand/Cement Replacement in Concrete. *Journal of Materials in Civil Engineering*, 21, p. 709-721.
- Tamanna, N. & Tuladhar, R. (2020). Sustainable Use of Recycled Glass Powder as Cement Replacement in Concrete. *The Open Waste Management Journal*, 13, p. 1-13.
- Tan, K. & Du, H. (2013). Use of waste glass as sand in mortar: Part I Fresh, mechanical and durability properties. *Cement and Concrete Composites*, 35, p. 109–117.
- Terro M, A.-G. I. (2003). Mechanical properties of concrete made with treated wastewater at ambient and elevated temperatures. *Kuwait Journal of Science & Engineering*, 30.

- Terro, M. J. (2006). Properties of concrete made with recycled crushed glass at elevated temperatures. *Building and Environment*, 41, p. 633-639.
- Tittareelli, F., Giouse, C. & Mobili, A. (2018). Recycled Glass as Aggregate for Architectural Mortars. *International Journal of Concrete Structures and Materials*, 12, 57.
- Topcu, İ. & Canbaz, M. (2004). Properties of concrete containing waste glass. *Cement and Concrete Research*, 34, p. 267-274.
- Ubeid, H. S., Hama, S. M. & Mahmoud, A. S. (2020). Mechanical Properties, Energy Impact Capacity and Bond Resistance of concrete incorporating waste glass powder. *IOP Conference Series: Materials Science and Engineering*, 745, 012111.
- Vinai, R. & Soutsos, M. (2019). Production of sodium silicate powder from waste glass cullet for alkali activation of alternative binders. *Cement and Concrete Research*, 116, p. 45-56.
- Wang, F., Gong, H., Jia, S., Chuan, B. & Zhang, C. (2014). Research on Waste Glass Concrete Basic Mechanical Property. *Applied Mechanics and Materials*, 711, p. 406-409.
- Wang, H.-Y. (2009). A study of the effects of LCD glass sand on the properties of concrete. *Waste Management*, 29, p. 335-341.
- Wang, W. C., Wang, S. Y. & Lin, C. H. (2017). Effect of High Temperature on Properties of Glass Concrete. *MATEC Web Conf.*, 138.
- Wilson, W., Soliman, N. A., Sorelli, L. & Tagnit-Hamou, A. (2019). Micro-chemomechanical features of ultra-high performance glass concrete (UHPGC). *Theoretical* and Applied Fracture Mechanics, 104, 102373.
- Yang C, H. R., Yeih W,S. (1995). Aggregate effect on elastic moduli of cement-based composite materials. *J Mar Sci Technol* 3, p. 5-10.
- Z.P. Bazant, W. J., and C.Meyer (1998). Micro fracturing caused by alkali-silica reaction of waste glass in concrete. Fracture Mechanics of Concrete Structures 3rd Int. Conf., FraMCoS – 3. Germany.
- Zeyad, A. M., Johari, M. A. M., Tayeh, B. A. & Yusuf, M. O. J. J. O. C. P. (2017). Pozzolanic reactivity of ultrafine palm oil fuel ash waste on strength and durability performances of high strength concrete. 144, p. 511-522.